

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Hasan, Rana; Mitra, Devashish; Ulubasoglu, Mehmet

# Working Paper Institutions and Policies for Growth and Poverty Reduction: The Role of Private Sector Development

ERD Working Paper Series, No. 82

**Provided in Cooperation with:** Asian Development Bank (ADB), Manila

*Suggested Citation:* Hasan, Rana; Mitra, Devashish; Ulubasoglu, Mehmet (2006) : Institutions and Policies for Growth and Poverty Reduction: The Role of Private Sector Development, ERD Working Paper Series, No. 82, Asian Development Bank (ADB), Manila, https://hdl.handle.net/11540/1878

This Version is available at: https://hdl.handle.net/10419/109289

#### Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.



http://creativecommons.org/licenses/by/3.0/igo

#### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



# WWW.ECONSTOR.EU



# ERD Working Paper ECONOMICS AND RESEARCH DEPARTMENT SERIES No.82

Institutions and Policies for Growth and Poverty Reduction: The Role of Private Sector Development

Rana Hasan, Devashish Mitra, and Mehmet Ulubasoglu

July 2006

ERD Working Paper No. 82

# INSTITUTIONS AND POLICIES FOR GROWTH AND POVERTY REDUCTION: THE ROLE OF PRIVATE SECTOR DEVELOPMENT

RANA HASAN, DEVASHISH MITRA, AND MEHMET ULUBASOGLU

July 2006

Rana Hasan is Senior Economist in the Development Indicators and Policy Research Division of the Economics and Research Department, Asian Development Bank; Devashish Mitra is Professor of Economics and Gerald B. and Daphna Cramer Professor of Global Affairs at the The Maxwell School of Citizenship & Public Affairs, Syracuse University; and Mehmet A. Ulubasoglu is Senior Lecturer in Economics at the School of Accounting, Economics and Finance, Deakin University. Asian Development Bank 6 ADB Avenue, Mandaluyong City 1550 Metro Manila, Philippines www.adb.org/economics

©2006 by Asian Development Bank July 2006 ISSN 1655-5252

The views expressed in this paper are those of the author(s) and do not necessarily reflect the views or policies of the Asian Development Bank.

# FOREWORD

The ERD Working Paper Series is a forum for ongoing and recently completed research and policy studies undertaken in the Asian Development Bank or on its behalf. The Series is a quick-disseminating, informal publication meant to stimulate discussion and elicit feedback. Papers published under this Series could subsequently be revised for publication as articles in professional journals or chapters in books.

# CONTENTS

Abstra	ct		vii
I.	Introd	luction	1
II.	Institu for Ec	itions and Policies and Why They Matter onomic Performance	3
	А. В.	Institutions Policies	3 6
III.	Survey	of Related Literature	7
	А. В.	Institutions, Policies, and Growth Institutions, Policies, and Poverty	7 10
IV.	Metho	dology	11
۷.	Data		13
VI.	Result	S	14
	А. В. С.	Summary Statistics and Broad Regional/Country Comparisons Growth Regressions Poverty Regressions	14 19 30
VII.	Conclu	ıding Remarks	35
	Select	ed References	36

# ABSTRACT

In this paper, we study the effects of institutions and policies on economic growth and poverty, paying close attention to institutions and policies that relate to the overall climate and regulations under which the private sector operates. We find that good governance, as measured by a strong commitment to the rule of law among other things, matters for poverty reduction largely through its effect on economic growth. Though not a panacea, less cumbersome regulations governing private sector operations, especially those pertaining to starting a business, can matter for both economic growth as well as poverty reduction more directly. While the impact of trade openness on growth is not clear, increases in trade shares have been associated with lower poverty. Furthermore, the size of the public sector has a strong negative effect on growth and can be bad for poverty alleviation. Finally, political freedom is not associated with either higher growth or lower poverty. Taken together, the evidence here suggests that the delivery of good governance and regulations, which facilitate the creation of new enterprises, is more relevant for growth and poverty reduction than the nature of the political system per se.

## I. INTRODUCTION

Despite having made significant progress in reducing poverty over the last several decades, 690 million people, constituting more than a fifth of Asia's population, live below the \$1-a-day poverty line (ADB 2004). On the basis of poverty lines more typically found in low-to-middle income countries as opposed to only low income countries—the \$2-a-day poverty line—an even more staggering number are poor in Asia: 60% of Asia's population or 1.9 billion people. Faced with such numbers, it is difficult to argue with the notion that the reduction of poverty is the central development challenge facing Asia, and indeed, the developing world at large.

What can policymakers do to reduce poverty? Many economists would argue that igniting economic growth and sustaining it is the surest and most sustainable way to fight poverty. Figure 1, which plots cross-country data on economic growth and poverty reduction, is consistent with this argument.<sup>1</sup> This figure shows, for example, that a 1% increase in growth has been associated on average with a 1.5% reduction in poverty. Moreover, episodes where poverty grew despite economic growth (quadrant 1) or where poverty declined despite economic contraction (quadrant 3) are clearly the minority. But Figure 1 also reveals that there is a great deal of variation in how much economic growth has reduced poverty across countries and even within countries over different periods of time. In statistical terms, the variation in economic growth can explain only around 45% of the variation in poverty reduction (ADB 2004).





<sup>&</sup>lt;sup>1</sup> Poverty reduction is based on changes in \$1-a-day poverty rates. The source of this figure is Key Indicators (ADB 2004).

These two "stylized facts" about growth and poverty linkages—that poverty reduction is closely associated with economic growth but that this association is by no means perfect—suggests two challenges for policymakers. First, what are the policies and institutions that can ignite and thereafter sustain growth. Second, how does one ensure that growth generates significant opportunities for the poor. In this paper, we discuss the evidence on both of these questions, paying special attention to policies and institutions that relate to the development of the private sector. In doing so, we are informed not only by existing cross-country studies examining the impact of policies and institutions on economic growth and poverty, we also carry out empirical analysis of our own using newly available data on poverty (World Bank-PovcalNet and ADB 2004) and indicators relating to the regulations under which the private sector operates (World Bank 2004).

Our focus on private sector development stems from the now widespread belief that market forces and private initiative provide a powerful basis for generating economic growth and development. This belief explains the importance private sector development is receiving in the strategies being adopted by both developing country policymakers as well as international development agencies. From the perspective of developing country policy making, for example, scores of countries have moved over the last 10 to 20 years to liberalize their trade and industrial policy regimes. These liberalizations have been underpinned by the belief that greater reliance on private agents—including not only large-scale manufacturing firms but also farmers and microentrepreneurs—to allocate resources on the basis of market signals would improve economic performance. Similarly, development agencies, including the Asian Development Bank (ADB) and the World Bank, have made support to private sector development a key component of their overall strategy for assisting less developed countries. ADB's private sector development strategy (ADB 2000), notes right at the outset that the "development of a strong and dynamic private sector is crucial to long-term, rapid economic growth." Long term, rapid economic growth is in turn seen as a "necessary condition for sustained poverty reduction."

The policies and institutions we focus on can be distinguished into four types. A first type relates to the various regulatory barriers faced by actual and potential private sector firms. These factors are measured in terms of variables available from the Doing Business database of the World Bank (World Bank 2004a). Examples include information on the ease of starting and closing businesses. A second type consists of broader outcome-based measures of overall policy orientations of the government that may affect the way businesses operate. These are captured by measures of openness to trade and the size of the public sector. While the former relates directly to how much domestic firms are exposed to foreign competition (via imports) and foreign markets (via exports), the latter captures quantitatively how important the private sector is in overall production. A third type relates to the institutions of governance. As is now increasingly appreciated, the most encouraging regulations (on paper) may not amount to much in terms of creating an enabling environment for the private sector if those regulations were enforced capriciously or by a predatory state machinery. Thus it is important to take into account the policy and institutional environment related to governance. A final set of variables relate to political institutions, in particular to political rights and civil liberties.

The results of our empirical work can be summarized as follows. We find that good governance as measured by a strong commitment to the rule of law, a competent and efficient government sector, and control of corruption matters for poverty reduction largely through its effect on economic growth. Though not a panacea, less cumbersome regulations governing private sector operations, especially those pertaining to starting a business, matter for both economic growth as well as poverty reduction more directly. While the impact of trade openness on growth is not clear, increases in trade shares have been associated with lower poverty. Furthermore, the size of the public sector has a strong negative effect on growth and can be bad for poverty alleviation. Finally, political freedom is not associated with either higher growth or lower poverty. Taken together, the evidence here seems to suggest that the delivery of good governance and regulations that facilitate the creation of new enterprises are more relevant for growth and poverty reduction than the nature of the political system through which such governance and regulations are delivered.

The paper is organized as follows. Section II provides a brief discussion on policies and institutions for private sector development and why these should matter for economic performance and thus economic development. Section III reviews the related literature. Section IV describes the methodology and Section V the data and their sources. Section VI provides an in-depth description and discussion of results. Section VII concludes.

# II. INSTITUTIONS AND POLICIES AND WHY THEY MATTER FOR ECONOMIC PERFORMANCE

The dividing line between institutions and policies is thin. Nevertheless, it is useful to try and distinguish between them for purposes of understanding their roles in improving economic performance. One basis for distinguishing between the two is that institutions encompass the formal and informal rules and customs within which individuals and firms operate; policies, on the other hand, refer to various strategies and measures a government adopts to achieve its goals and objectives within a country's institutional framework (Quibria 2002). Of course, policies can have a profound impact on a country's institutions, and this is ultimately what the objective of the efforts at policy reform in developing countries is about. In what follows, we discuss in more detail what institutions are, what they encompass, and why they are important to economic functioning and performance. We also briefly cover key policies that may have an important bearing on economic performance and on the process of institutional change itself.

# A. Institutions

The work of Douglas North has been an important influence on economists' thinking about institutions and their linkages with economic performance. North (1990) describes institutions as the "rules of the game" in a society. In his words, institutions encompass "humanly devised constraints that shape human interaction." These constraints can be formal or informal. The former would include constitutions, laws, and regulations governing politics and economics, while the latter would include conventions, customs, codes of behavior, and conduct. Formal rules and informal constraints together determine the incentives in human exchange: political, social, and economic. Importantly, how well the given institutions function depends on the nature of enforcement, that is, how costly it is to identify violations of the "rules of the game" and how severe the punishments for deviating from rules are.

Why are institutions important to economic performance? According to North, institutions have a profound influence on the incentive structure of a society. Countries are rich or poor

depending on whether their institutional constraints define a set of payoffs to political and economic activity that encourage productive activity. Put differently, when organizations, including firms, trade unions, political parties, business associations, etc. are engaged in what are unproductive activities it is because the institutional framework in which they operate provides them with an incentive to be unproductive. In developing countries the institutional framework "overwhelmingly favor activities that promote redistributive rather than productive activity, that create monopolies rather than competitive conditions, and that restrict opportunities rather than expand them. They seldom induce investment in education that increases productivity" (North 1990, 9). In this way institutions affect both the process of capital accumulation as well as the process of converting this capital into output.

If developing countries are poor because their current institutions provide a weak basis for providing the incentives that generate growth, what type of institutions should they acquire? And how can developing countries get there? Recent research has been in far more agreement on the first of these questions than the second.

Consider the first question. By and large, most economists today would assign a critical role to private incentives and initiatives in driving modern economic growth via the accumulation of capital and the conversion of that capital and labor into marketable output. Underlying this process of private accumulation, production, and exchange, however, has been a set of market-supporting institutions. The most fundamental among these have been the existence of secure and stable property rights and the rule of law. The importance of these was clearly noted and anticipated by Adam Smith in his *Wealth of Nations* (as cited in Rodrik, Subramanian, and Trebbi 2002, 1): "Commerce and manufactures can seldom flourish long in any state which does not enjoy a regular administration of justice, in which the people do not feel themselves secure in the possession of their property, in which the faith of contracts is not supported by law, and in which the authority of the state is not supposed to be regularly employed in enforcing the payment of debts from all those who are able to pay. Commerce and manufactures, in short, can seldom flourish in any state in which there is not a certain degree of confidence in the justice of government."

The experience of today's prosperous countries therefore suggests that stable property rights and the rule of law are prerequisites for a dynamic private sector to emerge. However, their experience also reveals that market economies require much more if they are to function efficiently and with at least a modicum of equity. As the experience of the 20<sup>th</sup> century has shown, market economies also need fiscal and monetary institutions that perform stabilizing functions, institutions that try to check market failures and regulate conduct in all sorts of markets including those dealing with goods and services, labor, and finance; and institutions that provide social insurance (Rodrik 1999).

It is difficult to imagine that today's developing countries, which are by and large transitioning to more fully market-based economic systems, will attain prosperity without these market-supporting institutions. The acquisition of these institutions is therefore part of their developmental challenge. However, it is far from clear how poor countries should go about acquiring these institutions (and as noted below, even how *urgently* they need all of them).

Rodrik (1999) discusses and contrasts two alternative approaches to the acquisition of institutions. According to the first, it is possible to "import [the] blueprint" or institutional design from the developed world. Following this approach the removal of price distortions and privatization of enterprises would be accompanied by a set of "governance" reforms that would include the

enactment of legal codes and legislations (often in line with those existing in the developed countries), establishment of an independent judiciary, etc.

The second approach, however, emphasizes that local circumstances will require in many cases a unique, context-specific institutional design. The usefulness and importance of such an approach can be understood and appreciated through the very real caution suggested by North (1994, 8): "economies that adopt the formal rules of another economy will have very different performance characteristics than the first economy because of different informal norms and enforcement. The implication is that transferring the formal political and economic rules of successful western economies to Third World and Eastern European economies is not a sufficient condition for good economic performance."

Consider the case of property rights and rule of law (investor perceptions of which are shown to be strongly associated with better economic performance as both our literature survey below as well as our own empirical work reveals). While strengthening property rights and rule of law in an economy where they have been weak or absent to begin with are very likely to be instrumental in getting entrepreneurs and investors to expand their investment activities, there is more than one way to establish stronger property rights and respect for the rule of law. An illustration from Rodrik (2004), contrasting Russia in the 1990s with People's Republic of China's (PRC) investment boom driven by the township and village enterprises in the 1980s (TVEs) reveals quite clearly the complexities involved in successful institutional change. In the 1990s Russia provided its citizens with a private property rights regime that was ostensibly enforced by an independent judiciary. However, surveys routinely revealed that investors consistently gave low marks to the country insofar as rule of law was concerned. Why? While Russia made the appropriate legislative changes, weaknesses in enforcement and the actual codes of conduct to which Russians were accustomed to probably diluted the effectiveness of these changes.

Put differently, context matters so that "institutional outcomes do not map into unique institutional design." Consider now the case of the PRC. The investment booms in the PRC's TVEs took place despite the nonexistence of a private property regime and an independent judiciary. Apparently, investors did believe their "property" to be safe. Drawing upon the work of Qian (2002), Rodrik (2004, 9) points to what may have served to provide secure property rights when none actually existed formally: with local governments typically the owners of the TVEs and private entrepreneurs as "effective" partners, "private entrepreneurs felt secure not because the government was prevented from expropriating them, but because, sharing in the profits, it had no interest to expropriate them."

Following the second approach's emphasis on local context, the general lesson is that "there is no unique, noncontext specific way of achieving desirable institutional outcomes." This, of course, makes it more difficult to give policy advice. But Rodrik argues that while good institutions are ultimately crucial for sustaining growth, igniting growth need not wait until large-scale institutional transformation has been generated. Instead, igniting growth entails the easier task of being able to identify and relax specific constraints that are holding back the private sector. Thus, in the case of the PRC in the late 1970s and 1980s, the main "binding" constraint to growth may have been the absence of market-oriented incentives. Doing away with the household responsibility system and allowing local governments to "own" the TVEs were innovations that gave entrepreneurs incentives to be productive within the existing overall institutional framework. More generally, the challenge for policy then becomes the search for responses that are appropriate for local conditions. This in turn requires an encouragement of experimentation—risks notwithstanding with reforms of different elements of the current institutional framework to see what types of reforms work and which ones do not.

# B. Policies

Policies can be thought of as the instruments by which governments can change the "rules of the game." Which policies have a particularly important bearing on economic performance? More to the point, what policies are holding back the private sector? Until very recently policy advice for generating growth and spurring the private sector tended to emphasize policy reforms focused on a relatively narrow range of areas. In particular, policy reforms have been targeted toward macroeconomic stabilization, price liberalizations, trade liberalizations, and the privatization of public sector enterprises.

Unfortunately, undertaking these policy reforms has not led to a big expansion in private sector activities; conversely, the private sectors of some countries that have avoided the above set of policy reforms have boomed (World Bank 2003). This has led analysts to look more closely at the full range of policies that influence the environment in which production takes place (as well as the institutional framework in which policies operate).

Rodrik (2003) contrasts two views on the policies and processes that can get "entrepreneurs excited about investing in the home economy." One view emphasizes cumbersome and misguided government regulations as the constraint to entrepreneurship and a vibrant private sector. The second view emphasizes market imperfections in developing countries. Accordingly, the government should not just get out of the way of the private sector. Instead it needs to find ways to crowd-in private investment.

According to the first view, government imposed imperfections, which include: "macroeconomic instability and high inflation, high government wages that distort the functioning of labor markets, a large tax burden, arbitrary regulations, burdensome licensing requirements, corruption" among others are holding back the private sector. Surveys of enterprises, including the investment climate studies (ICS) initiated by the World Bank and also being carried out by the ADB become a key tool in determining which aspects of government policy and regulations constrain the operations and growth of the private sector.

Per the second view, however, "economies can get stuck in the low-level equilibrium due to the nature of technology and markets, even when government policy does not penalize entrepreneurship." Rodrik points out that even though developing countries need not create new technologies they do need to adapt technologies that are new to them. This process of adaptation will usually require a certain amount of human capital internal to the individual and the firm contemplating use of the new technology. But crucially both the costs of adaptation as well as the returns from adaptation are often subject to externalities. For example, the costs of producing a new (for the developing countries) good may be uncertain. The uncertainties can deter investors. However, even one investor's entry into the new line of production would provide information on the costs. If the investor struggles, then this is a signal that the costs of production are high. If the investor, on the other hand, is successful then this is a signal that the cost of production are worth the effort. The point is that "entrepreneurs engaged in the cost discovery process incur private costs, but provide social benefits that can vastly exceed the anticipated profits." The end effect is that "the relevant learning is under produced in a decentralized equilibrium, with the consequences being that the economy fails to diversify into non-traditional, more advanced lines of activity. In such a scenario there will be policy interventions which would improve upon the decentralized outcome" (Rodrik 2003, 21).

Both of these views have merit. This is because both factors may be at work, even within the same country. Thus while learning externalities may be holding back certain types of investments, other investments could be constrained by unnecessarily burdensome regulation. Consider the wide range of regulatory policies that govern the rules and regulations on starting and closing a business. Cumbersome and/or costly regulations for starting a business could easily result in lower entry than otherwise; they could also lead to a lack of competition faced by existing firms. The result would not only be lower investment than otherwise, but also lower efficiency among incumbents protected from competition. Similarly, regulatory hurdles in closing a business would prevent firms that are currently inefficient from exiting the market. They may also deter entry by artificially raising the cost of exiting if market conditions ultimately prove to be too difficult for a firm.

A similar logic applies to labor market regulations. Labor market regulations encompass a number of issues, including the rules governing industrial relations and collective bargaining, hiring and firing of workers, minimum wage laws, health and safety regulations, and mandated benefits. These regulations can have significant implications for both efficiency as well as equity. For example, while restrictions on firing would protect the welfare of the employed, they may end up harming employment generation in general as firms respond to firing restrictions by reducing the number of workers they would otherwise hire. In an extreme case, strong restrictions to firing could even end up constraining investments, especially in labor-intensive activities as has often been claimed in the case of India (see, for example, Besley and Burgess 2003).

# **III. SURVEY OF RELATED LITERATURE**

# A. Institutions, Policies, and Growth

As may be alluded from the previous discussion, the work by North (1990) has generated a great deal of interest in the role of institutions in determining output and growth. Knack and Keefer (1995) looked at the impact of property rights on economic growth. In particular, they study the effects of contract enforceability and risk. While the variables are shown to have an impact on growth, controlling for them generates estimates that show faster conditional convergence to incomes levels in the United States. In a subsequent paper, Keefer and Knack (1997) look at how the convergence rate itself might be a function of indicators of institutional quality such as the rule of law, corruption, and the risk of expropriation and contract repudiation. They study this by interacting initial income with these institutional indicators on the right-hand side of a standard, cross-country growth regression.<sup>2</sup>

<sup>&</sup>lt;sup>2</sup> Another paper that pursues this line of research is Rodrik (1997), which looks at the role of institutions in explaining the stellar growth performance of East Asian economies prior to the Asian financial crisis. Almost all the variation in their growth performance is explained by initial income, initial education, and institutional quality. His measure of institutional quality is drawn from the work of Knack and Keefer (1995) and Easterly and Levine (1996).

Mauro (1995) focuses on the effect of corruption on growth and finds it to be negative in his cross-country regressions. He also looks at the effects of a subjective perceptions-based index of bureaucratic efficiency as well as that of political stability and finds them to be positive. Due to the possibility of reverse causation from growth to institutions, there is a potential endogeneity problem, which Mauro corrects by instrumenting corruption by an index of ethnolinguistic fractionalization.

Another important study is by Barro (1997) who studies a panel of around 100 countries from 1960 to 1990. Using controls such as the initial level of real per capita gross domestic product (GDP), initial schooling, life expectancy, fertility, government consumption, inflation, and terms of trade, growth is found to be increasing in the rule of law. Although Barro finds a weak effect of political freedom on growth, there is some indication of a nonlinear relation, an inverted Ushaped relationship between democracy and growth, with the growth-maximizing level of political freedom lying somewhere in between a pure democracy and pure dictatorship.

Using a new database consisting of 300 indicators of governance and creating six aggregates out of those, namely rule of law, corruption, political instability, voice and accountability, and government efficiency and regulatory burden, Kaufmann, Kraay, and Zoido-Lobaton (1999) show that there is a strong association between good governance and growth, with the causation running from the former to the latter. The instruments they use for their governance indicators are the ones used by Hall and Jones (1999).

While the above papers emphasize the effect of institutions on growth, later work has mainly emphasized the impact of institutions on income levels and not on growth rates. In support of this change in emphasis, Hall and Jones (1999) have argued that "levels capture the differences in long-run economic performance that are most directly relevant to welfare as measured by the consumption of goods and services." Also, in this context they point to the recent evidence on the transitional nature of growth rate differences across countries; the empirical questioning by Jones (1995) of the relevance of endogenous growth models; and the theoretical support from recent models that show the effect of policies on income levels and not on growth rates, and where countries in the long run differ in their income levels and not growth rates.

Hall and Jones (1999) look at how capital accumulation, productivity, and therefore output per worker are affected by social infrastructure. Social infrastructure here refers to institutional and policy variables that determine the economic environment determining capital accumulation, skill formation, invention, innovation and technology transfer. Their measure of social infrastructure is based on measures of corruption, expropriation risk, government repudiation of contracts, law and order, bureaucratic quality, and trade barriers. While output is made a function of social infrastructure in their estimation framework, they correct for endogeneity of the latter using instruments such as geographical variables, mainly distance from the equator and the extent to which modern European languages are spoken as first languages today, which captures European influences on institutions. Their study concludes that countries with better social infrastructure have higher levels of output per worker in the long run, have higher investment rates, and are more efficient at converting inputs to output.

Recently, a major advance in this literature has been made by Acemoglu, Johnson, and Robinson (2001) who looked at former European colonies to study the impact of institutions on per capita income levels. For these countries, they are able to use European settler mortality rates as instruments for institutions. In countries conquered by Europeans, whether they decided to permanently settle or not was determined by their ability to survive there (by their mortality rates). If they decided to settle in a country themselves, they adopted good institutions, while if they decided to rule from their home country, they put in place extractive institutions. Their decision to settle in a region, however, was a function of their mortality there. On the other hand, mortality rates of potential settlers, to begin with, can be viewed as a function of geographical variables. While even after instrumenting for institutions (expropriation risk that current and potential investors face), Acemoglu, Johnson, and Robinson find statistically significant effects of these variables on per capita income in the expected direction, the instrumentation completely takes away the effect of geographical variables on income.

Turning to trade policy, the effects of trade barriers on growth and income have been studied since the early 1990s. While Dollar (1992), Sachs and Warner (1995), and Edwards (1998) showed positive effects of trade on growth using different measures of openness, in many cases constructed from standard policy measures, these papers have been strongly criticized by Rodriguez and Rodrik (2001) for the problems with measures of trade openness and the econometric techniques used, as well as for the difficulty in establishing the direction of causality. While Rodriguez and Rodrik (2001) have criticized the measure of openness used by Sachs and Warner (1995) as capturing many aspects of the macroeconomic environment in addition to trade policy, Baldwin (2002) has recently defended that approach on the grounds that the other policy reforms captured in the measure, though not trade reforms per se, accompany most trade reforms sponsored by international institutions. Therefore, using such a measure tells us the value of the entire package of trade and accompanying reforms. Wacziarg and Welch (2003) have updated the Sachs-Warner data set and have again shown the benefit of such reforms in driving growth.

Just as in the case of the literature on the effect of institutions as explained above, the trade literature has also shifted focus to levels from growth rates. Frankel and Romer (1999) look at the effect of trade share in GDP on income levels across countries for the year 1985. They construct an instrument for the trade share by summing up the gravity-model driven, geography-based predicted values of bilateral trade flows across all trading partners. The variables used to predict bilateral trade flows include distance, country size variables such as land area and population, and dummies for whether the countries are landlocked or have a common border, etc. They find that their instrumental variables approach produces positive effects of trade on income levels that are greater than the estimates produced by ordinary least squares. Irwin and Tervio (2002) apply the Frankel-Romer approach to cross-country data from various periods in the 20<sup>th</sup> century to show that this trade–income relationship is indeed highly robust.

Building on two literatures, namely the one on institutions and incomes and the other on trade and incomes, Rodrik, Subramanian, and Trebbi (2002) looked at the simultaneous effects of institutions, geography, and trade on per capita income levels. They used a measure of property rights and the rule of law to capture institutions and the trade–GDP ratio to capture openness in trade, treating both as endogenous in their growth regressions. They used the instruments of Acemoglu, Johnson, and Robinson (2001) and Frankel and Romer (1999) to instrument institutions and trade openness, respectively, and separately. Rodrik, Subramanian and Trebbi (2002) find that "the quality of institutions trumps everything else." However, trade and institutions have positive effects on each other, so that the former affects incomes through the latter. Similarly, geography also affects institutions.

# B. Institutions, Policies, and Poverty

Moving from growth rates and incomes to poverty, we find that the literature on the determinants of poverty rates and changes (or reductions) in it is much smaller. Dollar and Kraay (2002), in a cross-country study of 92 countries over the last four decades, find that the growth rates of average incomes of people in the bottom quintile are no different from the growth rates of overall per capita incomes, with the former growth always positively associated with the latter. Thus the share of the bottom quintile of the population in overall income is fairly stable. Likewise, policies that promote overall growth advance growth in the incomes of the poor. These policies include trade openness, macroeconomic stability, moderate government size, financial development, and strong property rights and the rule of law.

One difficulty with interpreting the results of Dollar and Kraay is in their measure of poverty being a relative one. Due to their focus on incomes of the bottom quintile, it can be argued that their paper is more directly connected to the issue of inequality rather than absolute poverty per se. As a number of analysts have argued, insofar as developing countries are concerned, it is not so much relative poverty but absolute poverty that needs to be the focus of attention. Cross-country studies on absolute poverty are to our knowledge very limited. Ravallion (2001) studies the relationship between \$1-a-day poverty rates and economic growth. He finds that an increase in per capita income by 1% can reduce the proportion of people below the \$1-a-day poverty line by about 2.5% on average. This "poverty elasticity" varies across countries, depending on initial inequality. In other words, how close the poor are to the poverty line matters. Ravallion, however, does not examine the links between (absolute) poverty and policies. A recent paper that does so is that of Hasan, Quibria, and Kim (2003). A key finding of this paper is that measures of economic freedom are found to be closely linked to reductions in poverty.<sup>3</sup> A measure of political freedom, on the other hand, is not. Economic freedom indicators used by these authors include government size, price stability, freedom to trade with foreigners, and measures of political and civil liberties.

Moving to country-specific empirical work, a notable study is that of Ravallion and Datt (1999) on the determinants of poverty reduction across India's major states between 1960 and 1994. The study shows empirically how initial conditions—and thus initial inequalities—matter. Similar to the findings from cross-country comparisons of poverty–growth linkages, Ravallion and Datt find that the impact of a given amount of growth in nonfarm output on poverty reduction can vary considerably across India's states. For example, a 1% increase in nonagricultural state domestic product leads to a 1.2% decline in poverty rates in the states of Kerala and West Bengal versus only 0.3% decline in Bihar. The fact that growth of nonfarm output was also relatively meager in Bihar over the period under consideration exacerbated the poverty problem in Bihar.<sup>4</sup>

<sup>&</sup>lt;sup>3</sup> It may be noted that this paper derives estimates of absolute poverty by combining data on the distribution of per capita income (expenditure) with national account data on income (private consumption expenditure). Such an approach has been used by Bhalla (2002) and Sala-I-Martin (2002).

<sup>&</sup>lt;sup>4</sup> Ravallion and Datt then explore which factors "explain" this differential impact of nonfarm sector growth on poverty by state. Differences in initial conditions relating to rural development and human resources are found to be a key source of the interstate differential in poverty impacts of nonfarm output. The role played by initial literacy appears especially large. In particular, Ravallion and Datt find that more than half of the differential impact of nonfarm output on poverty rates is attributable to Kerala's much higher levels of initial literacy. Their results suggest that while the

## **IV. METHODOLOGY**

In this paper, we look at the determinants of growth and poverty focusing attention on institutions and policies, especially those relating to regulation of the private sector. Our variables on the right-hand side fall into two broad categories, namely institutional indicators and policy variables. Thus, our estimating equation for growth is the following:

$$g_{i} = \alpha_{0} + \alpha_{1} \cdot y_{i} + \alpha_{2} \cdot INV_{i} + \beta \cdot Inst_{i} + \gamma \cdot Policy_{i} + \varepsilon_{i}$$
(1)

where  $g_i$  stands for the average growth rate of per capita income during a decade (1990–1999),  $y_i$  stands for the logarithm of initial per capita income (at the beginning of the decade),  $INV_i$  stands for the average investment rate (investment as a fraction of overall GDP) for that decade,  $Inst_i$  is the decade average vector of institutional variables and  $Policy_i$  the vector of policy variables.<sup>5</sup> Our estimating equation for poverty is the following:

$$P_{i} = b_{0} + b_{1} y_{i} + B \cdot Inst_{i} + C \cdot Policy_{i} + v_{i}$$

$$\tag{2}$$

where  $P_i$  stands for the poverty rate, given by the under-\$2-a-day headcount ratio in 1999 and the other variables are as defined earlier.

The institutional variables include a measure of governance—the principle-component aggregate of measures of the rule of law, government efficiency, and corruption. These also include the "meta-institution" of democracy as captured by an index of political freedom. Why do we include a measure of political freedom? There is a strong presumption that the lack of property rights, high regulatory burden, and high levels of corruption, etc., result in high levels of rent-seeking activity and the exploitation of the rest of the society by the elite through the exercise of "public power for private benefit". Thus the poor could end up getting a very small share of an already small pie. In a democracy, however, it can be dangerous to keep a significant proportion of the population deprived and angry. Alternatively, political freedom could lead to populist policies that could end up reducing economic growth. We therefore look at the impact of political freedom and governance on the incidence of poverty.

Some have argued that institutions are endogenous to growth. However, in the new literature (Hall and Jones 1999; Rodrik, Subramanian, and Trebbi 2004) as explained in Section III, it is

transition from (low-wage) agriculture to (higher wage) nonfarm sectors may be key for the removal of poverty, making the transition is not easy or automatic for the poor. In other words, there are costs to be incurred on the part of a poor agricultural worker to making the transition. These costs are not only pecuniary ones but also nonpecuniary associated with investments in minimum levels of education, nutrition, and health so as to be able to work productively in the nonfarm sector.

<sup>&</sup>lt;sup>5</sup> The institutional and policy variables can affect the rate of growth through investment and other channels. Therefore, we run the above regression with and without INV as a regressor. This helps us somewhat in identifying the investment and noninvestment channels.

the income level and not the growth rate that depends on institutions, which in turn also depend on the income level itself. If we believe that institutions are slow to adjust, then using initial per capita income as a control should take care of the bulk of the endogeneity problem.

The policy variables capture key regulations governing the operations of the private sector. These include measures relating to the ease of starting a business, closing a business, and labor market policies that businesses must abide by. Our policy variables also include measures of government size and trade openness. We know that the public sector provides basic infrastructure and social services, and that, at one level, it can have strong complementarities with the private sector. However, too big a government can lead to inefficiency especially if it tries to operate in spheres that are normally meant for the private sector. Also, big governments can be captured by the elite and can promote rent seeking at the cost of real production. Trade, on the other hand, can generate efficiency through gains from specialization and exchange, as well as through the availability of larger varieties of final and intermediate goods. Thus, it can result in enhanced real incomes. While the theoretical foundations for enhanced real income levels through trade are strong, the growth effects are less clear cut. There is a vast and rich theoretical literature on trade and growth, but the results span the entire spectrum depending on the specifics of the assumptions made. Therefore, this is an empirical question that has received considerable attention recently but for which no conclusive answers have been provided so far.

Further, the size of the public sector and trade openness should also have an impact on poverty. As explained above, the size of the public sector should affect income and growth and thus, for a given distribution of income, have an impact on poverty. Additionally, the size of the public sector can also have an impact on the distribution of income. If the government believes in making society more egalitarian, it will use its machinery to provide social services for the relatively poorer sections of the society. On the other hand, the government can just be an instrument of the elite, which means that an increase in government size will worsen the distribution of income. We therefore empirically explore the relationship between poverty and the share of the government in overall GDP.

International trade can have an impact on poverty through its effects on the overall level of real income, the distribution of income, the sectoral composition of the economy, the relative rewards to different factors of production and the extent of urbanization. Again, trade affects poverty through both growth and distribution. As Bhagwati (2004, 53), focusing on the growth channel writes, "The scientific analysis of the effect of trade on poverty ... has centered on a two-step argument: that trade enhances growth, and that growth reduces poverty." Furthermore, Winters, McCulloch, and McKay (2004) argue that "although growth can be unequalizing, it has to be strongly so if it is to increase absolute poverty. This appears to be not the case either in general or for growth associated with freer trade." They argue that it is, in fact, the openness-growth link on which economists differ.

Other channels, according to Winters, McCulloch, and McKay through which trade can affect poverty are through its effects on overall economic stability, the creation and destruction of markets, output prices, wages, and employment. Also, how trade will affect wages of the poor will depend on the nature of factor endowments and comparative advantage. An important factor here is the extent of transitional unemployment caused by trade liberalization that gets concentrated on the poor. Finally, trade liberalization also does have an impact on the poor through its effects on government revenue.

# V. DATA

Our complete data set covers over a hundred countries, each with one data point (a decadal average unless otherwise stated) for each variable over the period 1990–1999. Fifteen countries are from developing Asia. Our data set covers a number of variables obtained from various sources. Growth is the average of annual percentage changes in real GDP per capita (base year 1995) within a decade. Initial income is the log real per capita income of 1990. INV is the share of investment (gross capital formation) in GDP; PUB is the share of general government final consumption expenditure in GDP; and trade/GDP is the sum of exports and imports of goods and services measured as a share of gross domestic product. We obtained these data from *World Development Indicators* (World Bank 2003).

Poverty is measured as the proportion of the population living on less than "\$2-a-day" and is primarily obtained from the PovcalNet database of the World Bank (World Bank 2004b).<sup>6</sup> For a few countries from developing Asia, however, we replaced the poverty estimates from PovcalNet with estimates from Key Indicators (ADB 2004). These countries are Cambodia, Lao PDR, Malaysia, Nepal, Pakistan, and Viet Nam.<sup>7</sup> While Chen and Ravallion (2004) describe the PovcalNet data in detail, the following points are useful to keep in mind. First, the PovcalNet database presents two types of poverty esitmates: "actual" and "synthetic." The former pertain to years in which nationally representative household income and/or expenditure surveys were carried out. The latter pertain to poverty estimates derived for common "reference" years.<sup>8</sup> We use poverty estimates from 1990 and 1999 reference years for the 76 developing countries that overlap across our sample and the PovcalNet database. While many of these reference-year estimates may be "synthetic" ones, including these in the analysis allows us to carry out a much richer analysis than would otherwise have been possible. Second, our choice of using the \$2-a-day poverty line, as opposed to the \$1-a-day poverty line popularly used in the international media, is due to the fact that it conforms better with poverty lines more typical in low-middle-income countries as opposed to low-income countries only. Additionally, using the \$2-a-day poverty line enables us to increase our sample size in a meaningful way. This is because while \$1-a-day poverty estimates exist for many middle-income countries, these estimates can be extremely low (close to 0) thereby reducing the variation in poverty rates in the way that could bias econometric results.

Political freedom is the simple average of the countries' political rights and civil liberties scores as given in the Freedom House (2003) report. The correlation between political rights and civil liberties is very high, so taking their simple average is expected to measure the overall political freedom in the countries.

We have also utilized a set of institutional variables capturing the state of "governance" obtained from Kaufman et al. (1999). These are measures of Rule of Law, Government Efficiency, and Control of Corruption. While we used these variables individually in the regressions, we also exercised

<sup>&</sup>lt;sup>6</sup> The term \$2-a-day is used for convenience. The actual figure is \$2.15 in 1993 PPP exchange rates for the consumption aggregate.

<sup>&</sup>lt;sup>7</sup> See ADB (2004) for more details.

<sup>&</sup>lt;sup>8</sup> Since most countries typically carry out the relevant household surveys every 3-5 years, it will be common to find that a survey is not carried out in one of the reference years. In all such cases, poverty estimates are obtained for reference years using the extrapolation methods of Chen and Ravallion (2004).

with their principal components aggregation, and we report the results with this variable. Additionally, because these data are available from 1996 onward, we used the 1996 values only, and in that way we limited the scope for reverse causation.

We also used a set of Doing Business variables obtained from World Bank (2004a). While there are five broad categories (entry regulations, labor regulations, contract enforcement, getting credit, and closing business) to measure business regulations, with several criteria within each category, we used some single measures. These are for the categories of entry regulations, closing business categories, and labor regulations, to capture the business environment, believing that the other categories are generally covered by our other governance variables. The original sources for these data are Djankov et al. (2002) and Djankov et al. (2003), respectively, for entry/starting a business, closing a business, and labor market regulations. In particular, for entry regulations we used the number of procedures for starting a business, cost of starting a business (as a percent of income per capita), and time/duration taken for starting a business (in days). As in the case of Kaufman et al. variables, we exercised with these single indicators as well as with their principal components aggregation. Regarding closing a business, we used the cost of closing a business (as a percent of the value of the estate) and time required for closing a business (in years), as both single and aggregated indicators. For labor market regulations, we worked with indexes of regulatory difficulties associated with hiring workers and firing workers (the values of both indexes ranging from 0 to 100), and the costs of firing workers (in terms of weekly wages).<sup>9</sup> Lastly, we aggregated these various Doing Business indicators using principal components to obtain aggregated Doing Business variables, and the results are also reported with these variables. The Doing Business variables-whether in aggregated form or in terms of their individual values-are again treated as belonging to the 1990s.

# **VI. RESULTS**

# A. Summary Statistics and Broad Regional/Country Comparisons

Tables 1 to 3 present summary statistics on the variables of interest in this paper. While Table 1 presents mean values by region for all but the Doing Business variables, Tables 2 and 3 present country-specific values for all variables pertinent to this paper's analysis including the Doing Business variables for not only ADB's developing member countries (DMCs) but also various other countries, developing and developed, to serve as points of comparison.

<sup>&</sup>lt;sup>9</sup> The difficulty of hiring index measures whether term contracts can only be used for temporary tasks; the maximum duration of term contracts; and the ratio of the mandated minimum wage to the average value added per working population. The difficulty of firing index is composed of various components that include whether regulations allow dismissal on the basis of worker redundancy, whether the firing of redundant workers require notification of labor union or labor ministry, whether their approval is required, etc. The cost of firing indicator measures the cost of advance notice requirements, severance payments, and penalties that may arise when firing a worker. All of these costs are totaled and expressed in terms of weekly wages. See World Bank (2004a).

Consider first the region-specific averages presented in Table 1. The DMCs are grouped into three categories:

- (i) DMC South Asia comprising Bangladesh, India, Nepal, and Pakistan
- (ii) DMC East and Southeast Asia (ESA) 3 comprising Republic of Korea; Singapore; and Taipei, China
- (iii) DMC East and Southeast Asia (ESA) 7 comprising Cambodia, People's Republic of China, Indonesia, Malaysia, Philippines, Thailand, and Viet Nam

The first three columns of Table 1 reveal that as of 1990 the highest poverty rates existed in South Asia, sub-Saharan Africa, and DMC ESA 7. As for changes in poverty, the largest declines took place in DMC ESA 7. South Asia also saw reasonable declines in poverty; however, this record is rendered a bit suspect due to the case of Pakistan for which national estimates of poverty reveal poverty to have increased slightly over the 1990s (see ADB 2004 for more details). In contrast, declines in poverty were on average meager for sub-Saharan Africa.

The experiences with poverty reduction follow closely the experiences with economic growth. Thus, the large declines in poverty experienced by ESA 7 have been accompanied by high growth per capita incomes-on average 4% per annum. Similarly, sub-Saharan Africa's meager reduction in poverty has been accompanied by a slight contraction of its economy over the 1990s. A clear outlier in this general pattern is the Middle East and North Africa region. In this region, despite positive though low economic growth, we find that poverty rates increased by 5 percentage points over the 1990s.

High growth in the ESA region has itself been accompanied by a high share of investment in GDP, low shares of government consumption expenditures in GDP, and a high degree of international integration whether measured in terms of trade shares or in terms of the SW-WH index of openness. By comparison, investment rates have been significantly lower in South Asia and the degree of international integration has been quite a bit less. Indeed, South Asia is less integrated than any other developing region in the world including sub-Saharan Africa. However, the share of government expenditures has been comparable to that of ESA.

South Asia also shows up performing poorly on our measure of governance (where lower values reflect worse governance). Only sub-Saharan Africa comes up with a poorer level of governance. On political freedom, however, South Asia does not perform too badly. In terms of the index of political freedom, whereby lower values reflect more political freedom, South Asia is found to have on average greater political freedom than ESA 7, sub-Saharan Africa, and Middle East and North Africa.

In summary, region-specific averages indicate that among developing regions where poverty is still a significant problem, ESA 7 has done the best job in reducing poverty. It has had the highest rates of economic growth, the highest rates of investment, and been quite open to trade. In terms of governance, however, its performance has been average while on political freedom it ranks low. At the same time, it should not be forgotten that poverty rates in ESA continued to be far higher than in three other major developing regions: Latin America, Middle East and North Africa, and Eastern Europe.

Since region-specific averages can mask important differences across countries we now turn to an examination of country-specific values taken by the variables of interest in this paper including

	Region
	BY
TABLE 1	<b>STATISTICS</b>
	SUMMARY

REGION	INITIAL POVERTY (1990)	FINAL POVERTY (1999)	CHANGE IN POVERTY (1990–99)	PER CAPITA INCOME GROWTH (1990–99)	INITIAL PER CAPITA INCOME (1990)	INVESTMENT SHARE (AVERAGE, 1990s)	GOVERNMENT EXPENDITURE (AVERAGE 1990s)	TRADE SHARE (AVERAGE, 1990s)	SW-WH (AVERAGE 1990s)	30VERNANCE (1996)	POLITTICAL FREEDOM (AVERAGE, 1990s)
DMC South Asia	76.8	68.9	-7.9	2.9	370.8	21.8	9.4	42.2	0.4	-1.0	3.7
DMC East & Southeast Asia 7	60.5	48.0	-12.5	4.0	1,110.0	28.6	10.0	83.2	0.8	-0.6	5.2
DMC East & Southeast Asia 3	n.a.	n.a.	n.a.	4.4	11,527.4	31.5	11.8	161.0	1.0	1.9	3.3
Eastern Europe	4.8	9.6	4.8	-1.0	4,152.7	23.5	17.6	84.7	0.7	-0.3	2.6
Latin America	30.7	29.1	-2.9	1.6	3,244.2	21.8	12.6	77.0	0.9	-0.7	2.6
Middle East & North Africa	17.0	22.3	5.3	1.6	4,850.1	23.1	20.5	85.3	0.6	-0.2	5.3
Sub-Saharan Africa	73.2	71.9	-1.1	-0.1	771.1	18.9	14.2	64.6	0.5	-1.3	4.6
Developed	N.A.	N.A.	N.A.	2.1	24,243.4	21.6	20.0	77.7	0.9	2.6	1.1

N.A. means not applicable/ not available.

				Summ	ABLE 2 ARY STATISTIC	S					
REGION	INITIAL POVERTY (1990)	FINAL POVERTY (1999)	CHANGE IN POVERTY (1990–99)	PER CAPITA INCOME GROWTH (1990–99)	INITIAL PER CAPITA INCOME (1990)	INVESTMENT SHARE (AVERAGE, 1990s)	GOVERNMENT EXPENDITURE (AVERAGE, 1990s)	TRADE SHARE (AVERAGE, 1990s)	SW-WH (AVERAGE, ( 1990s)	50VERNANCE (1996)	POLITICAL FREEDOM (AVERAGE, 1990s)
DMC											
Bangladesh	85.5	79.6	-5.9	3.0	278.2	19.1	4.5	25.7	0.4	-1.4	3.2
Cambodia	84.3	7.77	-6.6	1.7	239.6	15.8	8.5	55.4	n.a.	-1.9	5.9
PRC 71.5	50.0	-21.5	8.5	350.3	38.4	12.4	39.7	0.0	-0.5	6.9	
India 86.1	80.8	-5.3	3.7	323.9	23.1	11.4	23.0	0.0	-0.6	3.3	
Indonesia	70.9	55.4	-15.5	3.3	776.7	27.6	7.8	57.6	1.0	-0.8	5.7
Malaysia	11.4	9.6	-1.8	4.6	3104.0	36.3	12.1	178.1	1.0	1.0	4.6
Nepal	83.8	75.7	-8.1	2.4	188.0	22.9	8.8	50.0	0.9	-0.9	3.4
Pakistan	87.9	77.2	-10.7	1.4	448.2	18.7	12.5	36.4	0.0	-1.4	4.5
Philippines	54.9	48.1	-6.9	0.5	1090.9	22.4	11.3	82.4	1.0	-0.6	2.9
Singapore	n.a.	n.a.	n.a.	4.6	17620.4	35.5	9.5	328.6	1.0	3.6	4.7
South Korea	n.a.	n.a.	n.a.	5.2	7967.4	34.4	10.4	64.5	1.0	0.8	2.2
Sri Lanka	40.6	31.3	-9.3	4.0	615.9	24.9	9.9	75.9	0.9	-0.5	4.2
Taipei, China	n.a.	n.a.	n.a.	3.5	8994.3	24.6	15.4	90.0	1.0	1.4	2.9
Thailand	43.3	31.6	-11.7	4.2	1997.1	36.3	10.1	87.1	1.0	0.0	3.4
Viet Nam	87.4	63.6	-23.8	5.5	211.2	23.5	8.1	82.5	n.a.	-1.1	7.0
Latin America											
Argentina	n.a.	n.a.	n.a.	3.2	5775.8	17.8	10.0	18.7	0.9	-0.1	2.5
Brazil	32.3	23.7	-8.6	0.4	4079.1	20.8	18.4	18.0	0.9	-0.7	3.1
Chile 24.8	9.4	-15.4	4.9	3282.6	25.1	10.2	59.1	1.0	1.8	2.1	
Mexico	43.6	23.5	-20.1	1.7	3187.2	22.9	10.1	49.0	1.0	-0.8	3.8
Africa											
Botswana	55.7	48.1	-7.6	2.2	3067.7	29.4	27.3	94.0	1.0	0.5	2.0
Ethiopia	83.7	77.8	-6.0	1.2	100.3	14.4	13.5	30.6	0.4	-1.4	5.1
Ghana	74.3	78.5	4.2	1.6	345.9	20.7	12.5	62.7	1.0	-0.8	4.2
Kenya	61.2	64.9	3.8	-0.5	358.0	16.7	16.4	64.1	0.7	-1.8	5.8
Nigeria	86.6	91.2	4.7	0.2	258.5	19.8	12.9	79.8	0.0	-2.5	5.5
South Africa	31.8	33.5	1.7	-0.6	4112.6	14.6	19.6	44.3	0.9	0.4	2.8
Developed Countries											
France	n.a.	n.a.	n.a.	1.4	25966.7	19.8	23.6	44.6	1.0	2.4	1.5
Italy n.a.	n.a.	n.a.	1.3	18160.6	19.7	19.0	44.4	1.0	0.9	1.5	
Japan	n.a.	n.a.	n.a.	1.4	39955.4	29.2	14.6	18.2	1.0	2.0	1.6
United Kingdom	n.a.	n.a.	n.a.	1.9	18072.1	17.2	19.6	53.4	1.0	3.1	1.5
United States	n.a.	n.a.	n.a.	1.8	26140.6	18.2	15.6	22.5	1.0	2.8	1.0
n.a. means not applicable/ not	t available.										

TABLE 2

17 ERD WORKING PAPER SERIES No. 82

	START- PROCEDURES	START-TIME	START-COST (% OF INCOME PER	CLOSE-TIME	CLOSE-COST (% OF	LABOR- HIRING INDEX	LABOR- FIRING INDEX	FIRING COST
REGION	(NUMBERS)	(DAYS)	CAPITA)	(YEARS)	ESTATE)	(0-100)	(0-100)	(WEEKS)
DMC								
Bangladesh	8.0	35.0	91.0	4.0	8.0	11	20	47
Cambodia	11.0	94.0	480.1	n.a.	n.a.	33	30	39
PRC	12.0	41.0	14.5	2.4	18.0	11	40	90
India	11.0	89.0	49.5	10.0	8.0	33	90	79
Indonesia	12.0	151.0	130.7	6.0	18.0	61	70	157
Malaysia	9.0	30.0	25.1	2.3	18.0	0	10	74
Nepal	7.0	21.0	74.1	5.0	8.0	22	90	90
Pakistan	11.0	24.0	36.0	2.8	4.0	78	30	90
Philippines	11.0	50.0	19.5	5.6	38.0	22	40	90
Singapore	7.0	8.0	1.2	0.8	1.0	0	0	4
South Korea	12.0	22.0	17.7	1.5	4.0	11	30	90
Sri Lanka	8.0	50.0	10.7	2.2	18.0	0	80	108
Taipei,China	8.0	48.0	6.3	0.8	4.0	61	30	90
Thailand	8.0	33.0	6.7	2.6	38.0	67	20	47
Viet Nam	11.0	56.0	28.6	5.5	18.0	44	70	98
Latin America								
Argentina	15.0	32.0	15.7	2.8	18.0	44	30	94
Brazil	17.0	155.0	11.7	10.0	8.0	67	70	165
Chile	9.0	27.0	10.0	5.6	18.0	17	20	51
Mexico	8.0	58.0	16.7	1.8	18.0	67	90	83
Africa								
Botswana	11.0	108.0	11.3	2.2	18.0	0	40	19
Ethiopia	7.0	32.0	77.4	2.4	8.0	50	20	48
Ghana	12.0	85.0	87.5	1.9	18.0	11	50	25
Kenya	12.0	47.0	53.4	4.5	18.0	22	30	47
Nigeria	10.0	44.0	95.2	1.5	18.0	22	30	13
South Africa	9.0	38.0	9.1	2.0	18.0	56	60	38
Countries								
France	7.0	8.0	1.1	1.9	8.0	78	40	32
Italy	9.0	13.0	16.2	1.2	18.0	61	30	47
Japan	11.0	31.0	10.6	0.5	4.0	33	0	21
United Kingdom	6.0	18.0	0.9	1.0	6.0	11	10	25
United States	5.0	5.0	0.6	3.0	8.0	0	10	8

TABLE 3 BUSINESS REGULATIONS BY COUNTRY

n.a. means not applicable/ not available.

Source: World Bank (2004).

those pertaining to business regulation. An examination of changes in poverty by country, shown in column 3 of Table 2, reveals, for example, that while Latin America's performance in reducing poverty may have been lackluster on average over the 1990s, some individual Latin American countries did quite well in reducing poverty. For example, Mexico reduced poverty rates by 20 percentage points over the 1990s. This performance is comparable to the poverty reduction record of Vietnam and PRC. In contrast, the examination of changes in poverty in major African countries reveals that their performance has been much worse than that of the average country in that region. Similarly, it can be seen that ESA 7's high investment shares are driven mainly by the extraordinarily high values registered by PRC, Indonesia, Malaysia, and Thailand. On the other, hand, Philippines, Viet Nam, and especially Cambodia are marked by investment shares that are much lower than the average for this region. On governance, although South Asia registers a weak score, disaggregating the numbers by country, we see that the weak score is driven to a large degree by the very low scores for Bangladesh and Pakistan.

We now turn to the Doing Business measures of business regulation. What is interesting to note is the wide variance in the values taken by the various business regulation variables across the countries. Singapore is found to have the least cumbersome regulations governing entry and exit of businesses and labor issues in Asia. Singapore's regulations even compare favorably with the most developed countries in the world. By contrast, starting a business in Cambodia, India, and Indonesia is found to entail far more time; the relative costs for starting to business are also far higher, especially in Cambodia and Indonesia. Similarly, while it takes less than a year to close a business in Singapore it takes as long as 10 years to do so in India! This can represent a considerable wastage of capital—both physical as well as entrepreneurial—as the capital waits to get redeployed from a failing business to an alternative one. Next, while three countries in the region have few restrictions on the hiring of workers (for example, they do not disallow the usage of fixed term contracts for particular types of tasks) a number of other countries in the region, including Indonesia; Pakistan; Taipei, China; and Thailand put considerable restrictions on the types of contracts that can be used to hire workers. Interestingly, virtually every country, with the notable exception of Singapore from developing Asia, puts in place restrictions on firing workers. These restrictions are especially severe in India and Nepal followed by Indonesia, Sri Lanka, and Viet Nam. Finally, while firing workers entails a cost of four weeks of wages for firms in Singapore, this can be as high as 157 weeks in the case of Indonesia and 108 weeks in the case of Sri Lanka.

In what follows, we implement growth and poverty regressions involving the various variables we have discussed above including business regulation variables provided in the Doing Business database. In this way we are able to determine whether and how these variables are related to economic growth and poverty.

# B. Growth Regressions

Tables 4a–7 present the results of our growth regressions. Tables 4a-6b come in two versions. In Tables 4a, 5a, and 6a the Doing Business variables enter the regression equation expressed as principal components. In contrast, in Tables 4b, 5b, and 6b they appear in terms of the actual values taken by the individual regulatory measures, for example, as numbers of procedures, duration or time, and costs of starting a business, etc. While Tables 4a and 4b only introduce measures

of governance, political freedom, and the Doing Business measures as explanatory variables, Tables 5a–6b also introduce measures of openness and the size of public sector as elements of the overall, economywide policy setup. Initial income is included in every regression equation we run.<sup>10</sup>

Consider first the effect of governance on growth. As noted earlier, the variable governance is the principal component aggregation of the rule of law, government efficiency, and corruption variables of Kaufman et al. As is clearly seen in all tables, regardless of the controls used, governance is strongly significant at the 1% level with a robust positive sign. What is its quantitative impact on growth? Consider the first column of Table 4a where the coefficient of governance is 1.37. This implies that, holding all else constant, an improvement in governance from the value registered by Cambodia (the lowest in Asia) to the value registered by Singapore (not only the best in Asia but the best in the world) would lead to a 7.5 percentage point increase in the growth rate. This is clearly a dramatic effect. On the other hand, political freedom is mostly insignificant across specifications. In the few instances in which political freedom imply a more authoritarian regime, this suggests that controlling for initial income, governance, and other variables, greater political freedom has been weakly associated with lower economic growth. This would be consistent with recent work that has stressed the importance of good governance as distinct from political freedom being critical for driving economic growth (Glaeser et al. 2004).

As noted earlier, we introduced the Doing Business variables into the growth regressions in terms of two broad approaches. One approach aggregates individual variables in various combinations using principal component analysis. First, we aggregate the individual measures of business regulation relating to starting and closing a business and labor issues with the principal component method using the various individual measures in different combinations. Higher values of the principal component variables reflect more cumbersome regulations. Tables 4a, 5a, and 6a described these results. "Doing Business – 1" is the principal component aggregation of the number of procedures, costs, and the time for starting a business; and the costs and the time required for closing a business. The most notable result is that this "super-aggregated" variable is estimated with a negative and significant (at 10%) coefficient when governance is not in the regression; but in the presence of governance the coefficient becomes insignificant. Exactly the same result holds for "Doing Business – 2", which also includes in the principal component aggregation of "Doing Business – 1", indexes of difficulties in hiring and firing workers and a measure of the monetary costs of firing workers.

A less aggregated approach is also used for constructing principal component variables. "Start" is the principal component aggregation of the number of procedures, costs, and the time needed for starting a business; "close" is the principal component aggregation of the costs and the time required for closing a business; and "labor" is the principal component aggregation of indexes of difficulties in hiring and firing workers and the costs associated with firing workers. We employ these variables both separately and jointly in the regressions. All of them are estimated to be insignificant in the regressions, except one case of "close" where it is employed individually and

<sup>&</sup>lt;sup>10</sup> From all these tables, we see that there is some evidence of conditional convergence, depending on the controls used. In other words, in some cases the logarithm of the initial per capita income has a negative sign and is significant. The evidence is not so strong as our sample consists of only the time period 1990–1999 (due to the institutional data availability). The Asian financial crisis took place toward the end of this period and may have interfered with convergence effects among countries.

			PER CAF	PITA INCOME	E GROWTH			
Initial Income	-1.073 (4.06)***	0.080 (0.49)	0.009 (0.07)	-0.008 (0.06)	0.067 (0.48)	0.100 (0.82)	0.126 (1.05)	0.008 (0.05)
Governance	1.371 (5.27)***							
Political Freedom		-0.224 (1.07)						
Doing Business – 1			-0.318 (1.86)*					
Doing Business – 2				-0.300 (1.71)*				
Start					-0.289 (1.39)			-0.212 (0.96)
Close						-0.266 (1.62)		0.167 (0.96)
Labor							-0.217 (1.13)	-0.133 (0.68)
Constant	9.613 (4.70)***	1.259 (0.71)	1.103 (0.97)	1.239 (1.08)	0.626 (0.56)	0.406 (0.39)	0.184 (0.19)	1.226 (1.04)
Observations	116	131	112	112	115	112	115	112
Adjusted R-squared	0.24	0.02	0.02	0.03	0.02	0.01	0.01	0.01

TABLE 4A PER CAPITA INCOME GROWTH: INSTITUTIONS AND BUSINESS REGULATIONS I

Note: Robust t statistics in parentheses. \* significant at 10%; \*\* significant at 5%; \*\*\* significant at 1%.

		PER	CAPITA INC	OME GROWTH	I	
Initial Income	-0.099 (0.69)	-1.397 (5.43)***	0.105 (0.85)	-1.185 (4.52)***	0.114 (0.96)	-1.203 (4.26)***
Governance		1.919 (6.44)***		1.618 (5.41)***		1.662 (5.51)***
Political Freedom		0.306 (1.88)*		0.228 (1.30)		0.228 (1.37)
Start-Procedures	-0.070 (0.77)	0.167 (1.96)*				
Start-Time	0.005 (0.58)	0.008 (1.16)				
Start-Cost	-0.006 (5.46)***	-0.006 (7.23)***				
Close-Time			-0.041 (0.37)	0.020 (0.19)		
Close-Cost			-0.023 (1.42)	0.010 (0.38)		
Labor - Hiring					-0.016 (2.15)**	-0.001 (0.12)
Labor - Firing					-0.011 (0.97)	-0.015 (1.41)
Labor - Cost					0.009 (1.38)	0.014 (1.95)*
Constant	2.809 (1.75)*	9.429 (4.53)***	0.841 (0.69)	9.473 (4.67)***	0.750 (0.64)	9.520 (4.27)***
Observations	115	105	112	103	115	105
Adjusted R-squared	0.10	0.41	0.00	0.27	0.04	0.30

 TABLE 4B

 Per Capita Income Growth: Institutions and Business Regulations II

Note: Robust t statistics in parentheses.

\* significant at 10%; \*\* significant at 5%; \*\*\* significant at 1%.

				PER CAPII	A INCOME	GROWTH		
Initial Income	-0.925 (3.66)***	0.184 (1.15)	0.138 (0.97)	0.127 (0.88)	0.190 (1.34)	0.230 (1.73)*	0.269 (2.15)**	0.124 (0.84)
Government	-0.141	-0.111	-0.116	-0.123	-0.113	-0.108	-0.110	-0.122
Expenditure	(3.76)***	(2.98)***	(3.16)***	(3.30)***	(2.98)***	(3.01)***	(2.95)***	(3.15)***
Trade Share	0.007 (1.53)	0.011 (2.88)***	0.006 (1.56)	0.004 (0.97)	0.006 (1.72)*	0.007 (1.75)*	0.005 (1.16)	0.004 (1.01)
Governance	1.437 (5.33)***							
Political Freedom		-0.252 (1.23)						
Doing Business – 1			-0.366 (2.08)**					
Doing Business – 2				-0.370 (1.94)*				
Start					-0.350 (1.59)			-0.273 (1.18)
Close						-0.287 (1.73)*		-0.171 (0.98)
Labor							-0.251 (1.15)	-0.171 (0.77)
Constant	10.140 (4.70)***	1.482 (0.86)	1.491 (1.26)	1.828 (1.42)	0.959 (0.82)	0.603 (0.55)	0.416 (0.38)	1.778 (1.39)
Observations	115	130	112	112	115	112	115	112
Adjusted R-squared	0.31	0.08	0.06	0.07	0.06	0.05	0.05	0.05

	TABLE 5A			
PER CAPITA INCOME GROWTH: INSTITUTIONS	, BUSINESS REGULATIONS,	<b>O</b> PENNESS,	and <b>Government</b>	Expenditure I

Note: Robust t statistics in parentheses. \* significant at 10%; \*\* significant at 5%; \*\*\* significant at 1%.

		Р	ER CAPITA	INCOME GRO	WTH	
Initial Income	0.034 (0.24)	-1.243 (4.98)***	0.239 (1.75)*	-1.055 (4.30)***	0.240 (1.91)*	-1.050 (3.89)***
Government	-0.120	-0.136	-0.111	-0.150	-0.094	-0.123
Expenditure	(3.03)***	(3.68)***	(3.05)***	(3.85)***	(2.49)**	(3.09)***
Trade Share	0.005 (1.27)	-0.002 (0.40)	0.007 (1.79)*	-0.001 (0.22)	0.005 (1.06)	-0.001 (0.11)
Governance		1.984 (6.58)***		1.757 (5.82)***		1.731 (5.67)***
Political Freedom		0.322 (1.95)*		0.225 (1.23)		0.243 (1.37)
Start-Procedures	-0.105 (1.11)	0.126 (1.50)				
Start-Time	0.005 (0.60)	0.007 (1.11)				
Start-Cost	-0.006	-0.006	(5.73)***	(8.37)***		
Close-Time			-0.029 (0.25)	0.026 (0.26)		
Close-Cost			-0.027 (1.65)	0.005 (0.17)		
Labor - Hiring					-0.016 (2.02)**	-0.001 (0.14)
Labor - Firing					-0.008 (0.72)	-0.010 (1.11)
Labor - Cost					0.006 (0.83)	0.009 (1.12)
Constant	3.601 (2.12)**	10.824 (4.92)***	1.049 (0.82)	10.918 (5.22)***	0.935 (0.67)	10.389 (4.50)***
Observations	115	105	112	103	115	105
Adjusted R-squared	0.15	0.48	0.04	0.35	0.06	0.34

# TABLE 5B PER CAPITA INCOME GROWTH: INSTITUTIONS, BUSINESS REGULATIONS, OPENNESS, AND GOVERNMENT EXPENDITURE II

Note: Robust t statistics in parentheses.

\* significant at 10%; \*\* significant at 5%; \*\*\* significant at 1%.

			PER CAPI	FA INCOME G	ROWTH			
Initial Income	-0.859 (3.22)***	0.168 (0.76)	0.015 (0.09)	-0.017 (0.10)	0.040 (0.24)	0.090 (0.59)	0.079 (0.49)	-0.005 (0.03)
Government	-0.150	-0.083	-0.094	-0.102	-0.088	-0.082	-0.085	-0.101
Expenditure	(3.36)***	(1.92)*	(2.17)**	(2.37)**	(1.97)*	(1.93)*	(1.98)*	(2.29)**
SW-WH	0.317 (0.44)	1.634 (2.21)**	1.562 (1.77)*	1.570 (1.81)*	1.686 (2.01)**	1.610 (1.77)*	1.726 (2.02)**	1.588 (1.77)*
Governance	1.415 (5.15)***							
Political Freedom		-0.124 (0.47)						
Doing Business – 1			-0.298 (1.74)*					
Doing Business – 2				-0.333 (1.93)*				
Start					-0.304 (1.47)			-0.214 (0.98)
Close						-0.209 (1.21)		-0.090 (0.48)
Labor							-0.278 (1.48)	-0.219 (1.15)
Constant	9.955 (4.24)***	0.310 (0.13)	1.319 (1.16)	1.696 (1.45)	0.916 (0.84)	0.538 (0.53)	0.570 (0.59)	1.572 (1.33)
Observations	108	120	106	106	108	106	108	106
Adjusted R-squared	0.30	0.11	0.11	0.12	0.11	0.09	0.11	0.10

	TABLE 6A	
PER CAPITA INCOME GROWTH: INSTITUTIONS,	<b>BUSINESS REGULATIONS, OPENNESS,</b>	AND GOVERNMENT EXPENDITURE III

Note: Robust t statistics in parentheses. \* significant at 10%; \*\* significant at 5%; \*\*\* significant at 1%.

		PER	CAPITA INC	OME GROWTH	1	
Initial Income	-0.088 (0.51)	-1.135 (4.35)***	0.096 (0.62)	-0.951 (3.58)***	0.068 (0.41)	-0.914 (3.28)***
Government	-0.096	-0.130	-0.083	-0.150	-0.069	-0.118
Expenditure	(2.16)**	(3.05)***	(1.93)*	(3.34)***	(1.62)	(2.66)***
SW-WH	1.461 (1.73)*	0.706 (0.95)	1.598 (1.76)*	0.921 (1.17)	1.652 (1.89)*	0.990 (1.20)
Governance		1.843 (6.06)***		1.627 (5.42)***		1.553 (5.04)***
Political Freedom		0.393 (1.94)*		0.296 (1.44)		0.337 (1.57)
Start-Procedures	-0.093 (1.02)	0.118 (1.30)				
Start-Time	0.005 (0.63)	0.007 (1.09)				
Start-Cost	-0.006 (4.91)***	-0.006 (7.29)***				
Close-Time			-0.037 (0.29)	0.027 (0.26)		
Close-Cost			-0.017 (1.02)	0.011 (0.37)		
Labor - Hiring					-0.018 (2.36)**	-0.003 (0.45)
Labor - Firing					-0.007 (0.67)	-0.010 (1.09)
Labor - Cost					0.007 (0.90)	0.008 (1.09)
Constant	3.298 (1.95)*	9.115 (3.89)***	0.886 (0.71)	9.020 (3.86)***	1.016 (0.84)	8.257 (3.40)***
Observations	108	100	106	99	108	100
Adjusted R-squared	0.19	0.48	0.09	0.36	0.13	0.35

# TABLE 6B PER CAPITA INCOME GROWTH: INSTITUTIONS, BUSINESS REGULATIONS, OPENNESS, AND GOVERNMENT EXPENDITURE IV

Note: Robust t statistics in parentheses.

\* significant at 10%; \*\* significant at 5%; \*\*\* significant at 1%.

				PER CAPITA	INCOME	GROWTH		
Initial Income	-1.079 (4.01)***	-1.095 (4.08)***	-0.099 (0.61)	-1.187 (4.71)***	0.048 (0.33)	-1.076 (3.93)***	0.015 (0.09)	-1.123 (4.20)***
Expenditure	-0.140 (3.20)***	-0.141 (3.18)***	-0.109 (2.69)***	-0.128 (2.99)***	-0.098 (2.54)**	-0.144 (3.26)***	-0.085 (2.11)**	-0.109 (2.51)**
SW-WH	0.834 (1.16)	0.820 (1.11)	1.541 (1.87)*	0.683 (0.91)	1.694 (1.96)*	0.841 (1.15)	1.652 (1.97)*	0.795 (1.07)
Governance	(5.12)***	(5.22)***		(6.04)***		(5.21)***		(5.33)***
Political Freedom	0.164 (0.83)	0.148		0.293		0.169 (0.84)		0.187 (0.90)
Doing Business – 1	0.160 (0.86)	(0110)		()				(000)
Doing Business – 2		0.103 (0.59)						
Start-Procedures		<b>、</b>	-0.091	0.114				
Start-Time			0.003	0.007 (1.05)				
Start-Cost			-0.004 (3.81)***	-0.005				
lose-Time			()	()	-0.032	0.011		
Close-Cost					-0.004	0.014		
Labor - Hiring					(0.25)	(0.59)	-0.009	0.003
Labor - Firing							-0.009	(0.40) -0.014 (1.72)*
Labor - Cost							0.005	0.010
Investment Share	0.121 (2.57)**	0.122 (2.65)***	0.104 (2.94)***	0.075 (1.91)*	0.128 (3.43)***	0.120 (2.53)**	0.119	0.136 (3.01)***
Constant	7.927	8.087 (3.31)***	1.193	8.152 (3.41)***	-1.651	7.718	-1.120	7.227
Observations	99	99	108	100	106	99	108	100
Adjusted K-squared	0.43	0.43	0.25	0.50	0.20	0.42	0.22	0.44

TABLE 7 PER CAPITA INCOME GROWTH: INSTITUTIONS, POLICIES, AND INVESTMENT SHARES

Note: Robust t statistics in parentheses. \* significant at 10%; \*\* significant at 5%; \*\*\* significant at 1%.

the controls are initial income, government expenditure, and share of trade in GDP (Table 5a). In this case, it has a negative sign significant at the 10% level. In other words, regulations that make it more difficult to close a business are associated with lower growth.

A second approach to introducing the Doing Business variables is to use the actual values taken by the various measures of regulation. It is useful to recall that as the values of these variables rise, regulations become more cumbersome. The key findings are as follows. Most notably, starting cost is robustly significant at the 1% level with a negative sign, both with and without the governance and the political freedom variables in the regressions. The coefficient is estimated consistently to be around -0.005. This implies that holding all else constant, a decline in starting costs from the levels recorded for Cambodia (by far the highest in Asia) to those recorded for the United States (among the lowest in the world) would lead to a 2.4 percentage point increase in growth rates. Starting procedures, however, has unxpectedly a positive and significant (at 10%) sign in one case (Table 4b) so that an increase in starting procedures is associated with an increase in growth rates. However, the impact of this variable is not very robust. When government expenditure and openness are controlled for (as in Tables 5b and 6b), the impact of starting procedures turns to be insignificant. Starting time is in all cases estimated as insignificant.

Closing time and closing costs are estimated insignificant in all cases as well. However, firing costs have the same result as starting procedures: a positive and significant (at 10%) sign in one case (Table 4b) so that the higher the firing costs, the higher the growth. But again, when government expenditure and trade policy are controlled for, this sign turns to be insignificant (as in Tables 5b and 6b).

Cumbersome labor hiring regulations have a negative and significant impact (at the 5% level) on growth in the absence of the governance and political freedom variables as controls.<sup>11</sup> The coefficient on labor hiring implies that, holding all else constant, moving from the level of difficulties in hiring regulations seen in Pakistan (the most difficult in Asia) to those in Singapore or Sri Lanka (the least difficult) would lead to an increase in growth of about 1.25–1.4 percentage points. Interestingly, difficulties in firing labor always appear with a negative sign, implying that the greater regulatory barriers to firing workers are associated with lower economic growth. However, the effect is typically found to have an insignificant impact on growth, the exception being when governance, political freedom, and investment variables are jointly controlled for in the regression. In this case regulatory restrictions on firing workers have a negative and significant coefficient at 10%.

The results with the Doing Business variables imply that there is some evidence that private sector regulations have an impact on growth. But the results also suggest that it is not individual components of the set of regulations but the overall package that is more important. In other words, there could be several alternative combinations of various regulations that might result in a given growth rate, and this results in the significance of the overall index and not the individual components. Also, it is only through better governance that better packages of regulations can be delivered, and therefore controlling for the former takes away the effect of the latter.

From our regressions, we cannot really say that openness as measured by the share of trade in GDP has any systematic effect on the growth rate of an economy. In most of the regressions, this variable is insignificant. The impact of openness, however, is often positive when we use the

<sup>&</sup>lt;sup>11</sup> This suggests that the influence of labor hiring on growth works through these variables.

Sachs-Warner index of openness updated by Wacziarg and Welch (SWWH). When this index is significant, the coefficient is estimated to be between 1.5–1.7. This implies that completely open regimes—such as that of Singapore (SWWH index = 1)—have, on average, growth rates that are 1.5–1.7 percentage points higher than those of closed regimes such as that of India (SWWH index = 0). However, very interestingly, the significance of SWWH is conditional upon the absence of the governance variable in the regression. There are two possible reasons for this conditional significance of the SWWH dummy. Firstly, the opening of the trade regime, as shown by the SWWH dummy, captures more than trade liberalization; it involves a higher scale policy reform and institutional changes.<sup>12</sup> Secondly, the governance variable itself could capture trade policy if good trade policy itself is an integral part of good governance. A possible implication of this conditional significance of the trade policy variable is that all well-governed economies are open to trade but not all open economies are well governed; and that openness, in the absence of good governance, is irrelevant for economic growth.

Lastly, we consider the size of public sector, as measured by the share of government expenditure in GDP, and trade policy variables. In almost all our growth regressions, government expenditure is negative and significant. This means that the larger the size of the public sector relative to the overall economy, the lower is the growth rate. While the public sector provides infrastructure essential to the proper functioning of the private sector, an increase in its size could also portend a reduction in the more dynamic component of the economy, namely the private sector. Also, the public sector may breed corruption and inefficiency, and increase the incentives for unproductive activities such as lobbying. We see from our regressions that a percentage point increase in the share of the public sector is generally associated with a 0.1–0.15 percentage point reduction in the growth rate.

As one can see, our regressions, except the ones presented in Table 7, do not include INV (investment as a share of GDP) as a right-hand side variable. The reason is that our institutional, policy, and governance variables may affect growth through their effect on investment as well as independently of it. In the former case, we would like to allow the parameter estimates of the policy and institutional variables to capture their full role in the determination of growth through all possible mechanisms. But as we clearly see in Table 7, there is no drastic change in the impact of policy and institutional variable on growth when INV is controlled for in the regressions as compared to the results reported in Table 6b. The results for government expenditure, SWWH, governance, and starting costs are gualitatively the same. There are only three changes over previous results. First, although countries that lack political freedom continue to tend toward having higher growth rates, this effect is not statistically significant in any specification. Similarly, restrictions on hiring workers now do not show up with a significant negative impact on growth. Both suggest that the impact of these variables on growth work through their (positive) effect on investment rates. Interestingly, however, there is some evidence now that restrictions on labor firing can have a negative and significant impact on growth (last column). The coefficient of -0.014 implies that holding all else constant, moving from a regime such as that of India (difficulty of firing worker index = 90) to that of Singapore (index equal 0) would add 1.26 percentage points to growth rates.

<sup>&</sup>lt;sup>12</sup> Rodriguez and Rodrik (2001) note that the SW dummy is a better measure of institutional change than being a trade policy indicator.

As for the impact of investment on growth, as can be seen from examining Table 7, investment has a positive and robust effect on growth. This is consistent with the old Harrod-Domar type as well as Mahalanobis type growth models. It is also consistent with the prediction of recent endogenous growth theories. Additionally, Easterly and Levine (1997) find that among nearly 50 variables considered in growth regressions, the most robust effect is that of investment. A high rate of investment helps build up the economy's future capacity to produce, and hence results in higher growth. We see from our regressions reported in Table 7 that a 15 percentage point increase in the investment rate—the difference in investment rates across the PRC and India—can bring about a 1.2–2.1 percentage point increase in the growth rate depending on the particular specification employed.

In sum, a lot of the cross-country differences in growth can be explained by differences in governance quality, investment, and size of the public sector. Since good governance should lead to good policies, and we do not see good (bad) policies when the quality of governance is poor (good), policies do not exhibit an independent impact on growth. Additionally, policies will be effective for growth as long as the institutional framework allows them to do so. However, investment and public sector size do seem to matter for growth independently of institutions, as they may not entirely be a function of the nature, quality, and style of governance. While savings, and therefore investment rates, also depend on rates of time preference and intergenerational altruism of households, the size of the public sector depends on the government's inequality aversion and its concern for unemployment reduction. In other words, it is possible to have good governments both to the left and right of center, and they may resolve the equity–efficiency trade-offs in different ways.

# C. Poverty Regressions

Tables 8 through 10 are our poverty regressions. Table 8 includes as regressors only our institutional variables (governance and political freedom) and the Doing Business variables introduced in terms of the two approaches noted above (that is, in terms of various principal component aggregations as well as values taken up by the individual measures). Table 9 introduces the share of government expenditure as well as trade shares as a measure of openness, while Table 10 uses the SWWH index as a measure of openness in place of trade shares. Initial income is included in all specifications.<sup>13</sup>

A notable result and one that is very different from the growth regressions above is that the poverty regressions mostly fail to produce significant estimates for the governance variable. However, it should be mentioned that the sign of the governance variable is always negative, indicating that better governance has been associated with lower poverty levels. Moreover, in the one case where governance does show up with a statistically significant impact—the case where government expenditure, trade policy as measured by the SWWH index, and closing time and costs are jointly controlled for (Table 10)—the "economic" impact of governance can be very large. The size of

<sup>&</sup>lt;sup>13</sup> As we can see, a higher initial per capita income is negatively correlated with the poverty rate. A 1% increase in the initial per capita income can lead to a reduction in the poverty rate from anywhere between 20 and 23 percentage points. While these are pretty extreme numbers, they are a reflection of a significant bunching of large groups of the population around the \$2-a-day poverty line.

	\$2-A-DAY POVERTY									
Initial Income	-22.412 (11.13)***	-22.799 (10.75)***	-20.936 (11.77)***	-23.500 (12.06)***	-22.235 (15.96)***	-22.553 (15.61)***	-22.151 (15.53)***	-20.607 (11.92)***	-23.014 (16.50)***	-23.277 (16.59)***
Governance	-2.530	-2.630	-3.347	~ /	( )		<b>、</b> ,	~ /	~ /	( )
Political Freedom	-2.131	-2.457		-1.486 (0.87)						
Doing Business – 1	3.206	(100)		(0.07)	3.568 (2.52)**					
Doing Business – 2	(2113)	1.518			(2:52)	1.619				
Start		(1114)				(1.57)	4.514			
Close							1.021			
Labor							-1.287			
Start-Procedures							(0.52)	0.441		
Start-Time								0.061		
Start-Cost								0.051		
Close-Time								(2.24)	-0.132	
Close-Cost									(0.14) 0.234 (1.59)	
Labor - Hiring									(1.55)	0.048
Labor - Firing										0.071
Labor - Cost										-0.096
Constant	194.584	205.776	182.997	210.387	194.584	197.261	194.179	172.896	197.918	204.321
Observations	73	68	72	80	73	73	73	76	73	76
Adjusted R-squared	0.75	0.72	0.70	0.69	0.75	0.73	0.75	0.77	0.73	0.75

		TABLE 8	3		
\$2-A-DAY	<b>POVERTY:</b>	<b>I</b> NSTITUTIONS	AND	<b>B</b> USINESS	REGULATIONS

Note: Robust t statistics in parentheses. \* significant at 10%; \*\* significant at 5%; \*\*\* significant at 1%.

				\$2-A-DAY P	OVERTY			
Initial Income	-22.652 (10.05)***	-23.007 (9.76)***	-20.418 (11.22)***	-21.598 (8.83)***	-22.740 (15.22)***	-22.970 (10.14)***	-22.496 (14.68)***	-22.494 (9.95)***
Government Expenditure	0.808 (1.79)*	0.808 (1.76)*	0.706 (1.73)*	0.681 (1.56)	0.863 (2.02)**	0.880 (1.81)*	0.476 (1.04)	0.469 (0.94)
Trade Share	-0.087	-0.091	-0.120	-0.102	-0.134	-0.122	-0.141	-0.126
Governance	-2.630	-2.686	(1.95)	-1.447	(2.23)	-3.282	(1.90)	-3.261
Political Freedom	(1.04) -2.161 (1.35)	(1.01) -2.489 (1.49)		(0.49) -2.341 (1.55)		(1.20) -2.387 (1.40)		(1.23) -2.445 (1.39)
Doing Business – 1	3.182 (1.96)*	( ••••)				(,		
Doing Business – 2		1.497 (0.99)						
Start-Procedures			0.222	0.300				
Start-Time			0.065	0.073				
Start-Cost			0.051	0.044				
Close-Time			(2.71)	(1.55)	-0.237	-0.147		
Close-Cost					0.264	0.269		
Labor - Hiring					(1.00)	(1.00)	0.020	-0.017
Labor - Firing							0.051	0.088
Labor - Cost							-0.095	-0.083
Constant	197.792	202.071	171.851	185.776	192.696	199.031	203.738	(1.01) 207.937 (11.25)***
Observations Adjusted R-squared	68 0.74	68 0.73	76 0.77	70 0.76	73 0.75	68 0.73	76 0.76	70 0.74

 TABLE 9

 \$2-a-Day Poverty: Institutions and Policies (Trade Shares)

Note: Robust t statistics in parentheses.

\* significant at 10%; \*\* significant at 5%; \*\*\* significant at 1%.

				\$2-A-DAY	POVERTY			
Initial Income	-22.416	-22.902	-19.726	-20.743	-23.267	-22.782	-23.021	-22.685
Government	0.326	(9.49)****	0.016	0.093	0.203	0.347	-0.154	-0.029
Expenditure	(0.61)	(0.65)	(0.03)	(0.17)	(0.39)	(0.63)	(0.29)	(0.05)
SW-WH	-3.395	-2.839	-5.658	-5.643	-3.827	-3.188	-3.680 (0.64)	-1.795
Governance	-3.125	-3.189	(1.02)	-1.558	(0.03)	-4.497	(0.01)	-3.910
Political Freedom	-1.902	-2.308		-2.293 (-1.53)		-2.040		(1.50) -2.173 (1.21)
Doing Business – 1	3.764	(1.55)		(1.55)		( 1122)		(1121)
Doing Business – 2	()	2.163 (1.61)						
Start-Procedures		· · ·	0.441 (0.67)	0.449 (0.55)				
Start-Time			0.061 (1.16)	0.068 (1.12)				
Start-Cost			0.069	0.064				
Close-Time			()	()	0.309	0.508		
Close-Cost					0.292	0.319		
Labor - Hiring					()	()	0.058	0.022
Labor - Firing							0.065	0.096
Labor - Cost							-0.087	-0.078
Constant	197.576	201.863	168.842	181.993 (9 49)***	196.606 (16.87)***	193.533	206.184	204.926
Observations	66	66	73	67	71	66	73	67
Adjusted R-squared	0.74	0.73	0.78	0.76	0.75	0.73	0.75	0.73

TABLE 10 \$2-A-DAY POVERTY: INSTITUTIONS AND POLICIES (TRADE POLICY)

Note: Robust t statistics in parentheses. \* significant at 10%; \*\* significant at 5%; \*\*\* significant at 1%.

the estimated coefficient on governance implies that India's poverty rate would be 18.9 percentage points lower if its governance was as good as that of Singapore (that is, a value of 3.6 rather than -0.6 for our measure of governance) while holding all else constant. As for political freedom, this variable is estimated to be insignificant in all cases, a case generally in concurrence with the growth regressions.

Interestingly, "Doing Business – 1" is estimated to be significant in all cases at levels varying between 5–10%. The average of the coefficient is 3.4. "Doing Business – 2", however, is estimated as insignificant in all regressions. Recall that this variable includes "start", "close" and "labor" in its aggregation, suggesting that something regarding the labor regulations is working against the Doing Business – 1 variable. This can be confirmed by examining the business regulation variables on a more disaggregated basis. Indeed, as suggested by column 7 of Table 8, which reports the results from principal component aggregations of Doing Business variables by separate "start", "close" and "labor" groupings, the significant impact of Doing Business – 1 on poverty is driven by regulations having to do with starting businesses. Regulations associated with closing businesses also matter in the same direction, that is, an increase in regulations is associated with greater poverty, albeit not significantly so. However, increased regulations relating to labor are associated with lower poverty (though not significantly so).

To get a better sense of which specific types of regulations affect poverty, we can turn to the remainder of the results that are based on including the Doing Business variables in terms of the values taken by individual measures of regulation. Starting costs are estimated to have a robust adverse impact on poverty in all cases, significant at 1% level. The coefficient centers on 0.055. To get a sense of the quantitative impact of this variable on poverty we can note that a reduction in starting costs from the levels registered in Cambodia (by far the highest in Asia and perhaps the world) to levels registered in the United States would, holding all else constant, lead to a decline in poverty rates of 26.4 percentage points. Taking less extreme values, we can note that a reduction in starting costs from the levels registered in Indonesia to levels registered by its Southeast Asian neighbor, Thailand, would lead to a reduction in poverty rates of 6.8 percentage points. Clearly, this is not a trivial decline. In contrast to starting costs, starting procedures and starting time are estimated to be insignificant in affecting poverty.

As for closing regulations, while the estimate for closing time is always insignificant closing cost is estimated to have some impact on poverty, with a positive sign that is significant between 5-10% levels. Focusing on the statistically significant estimates, a decline in closing costs from levels registered in the Philippines, where among Asian countries they are the highest, to a level registered by Pakistan, among the lowest in Asia, would lead to a decline of around 8.8–9.9 percentage points in poverty rates holding all else constant. Lowering closing costs to levels registered in Singapore would lead to even lower poverty rates.

Among labor regulations variables, firing costs enter with a negative sign, which in some cases are significant at the 5–10% levels. This is interesting, as it implies that an increase in firing costs decreases poverty. Thus, while higher firing costs may be a detriment to investments, our results are consistent with the notion that higher firing costs lead firms to either fire workers in lesser numbers or that the various payments that firms must make to workers they are firing helps them from falling into poverty. On the other hand, difficulties with hiring and firing regulations seem not to have any impact on poverty. Overall, except for the firing costs results, the results with disaggregated Doing Business variables are in line with the results for the growth regressions.

The impact of openness on poverty depends crucially on which measure of openness is used. In striking contrast to the results from the growth regressions, we find that larger trade shares are often associated with lower poverty while a higher value of the SWWH index of openness is always insignificantly related to poverty. These results suggest that while a more open trade *policy* may not lead to significantly lower poverty, countries that have managed to increase trading volumes relative to their GDP have often done so with benefits flowing to the poor. This is not as weak a result as some may claim. For example, media reports sometimes claim that trade has benefited countries at the expense of their poor. Our results here do not lend support to such reports.

Interestingly, the particular measure of openness used also has an important bearing on the estimated impact of government expenditures on poverty. When trade shares are used as the measure of openness, increased government expenditures show up with a positive and significant impact on poverty. In other words, a larger public sector (as proxied by government expenditures) is significantly associated with higher poverty. However, moving to the SWWH index of openness completely wipes out the statistical significance on the coefficient of government expenditures. This is understandable, as large government expenditures are a sign of fiscal irresponsibility and the SWWH captures macroeconomic/fiscal reforms that accompany most trade reforms (see Rodriguez and Rodrik 2001).

In sum, initial income, "Doing-Business – 1" and starting costs turn out to be the important variables that matter for poverty. In some contrast to the results for the growth regressions, this suggests more room for policies that could fuel private sector activities in the economy. The fact that governance and political freedom variables are mostly insignificant suggests that governments of the same quality can resolve the equity–efficiency trade-offs in different ways. Noting that governance is always significant in growth regressions, this further suggests that the actions to be taken to reduce poverty are quicker than those to increase steady-state growth, which requires institutional modification.

# VII. CONCLUDING REMARKS

In this paper, we have studied the role of institutions and policies in driving economic growth and explaining poverty, paying close attention to institutions and policies that relate to the overall climate and regulations under which the private sector operates. We find that good governance as measured by a strong commitment to the rule of law among other things, matters for poverty reduction largely through its effect on economic growth. While the impact of trade openness on growth is not clear, increases in trade shares have been associated with lower poverty. Furthermore, the size of the public sector has a strong negative effect on growth and can be bad for poverty alleviation. While a large public sector can be an asset to the rest of the economy in terms of the infrastructure and social services it provides, it also means that the economy has a relatively smaller private sector, usually the more dynamic sector. We interpret our regressions to show that in most countries, growth is driven by the dynamism of the private sector, which in turn depends importantly on the quality of governance.

Less cumbersome regulations governing private sector operations, especially those pertaining to starting a business, matter for both economic growth as well as poverty reduction more directly. However, it is important to point out that eliminating cumbersome regulations is unlikely to serve as a panacea. There are no doubt regulatory practices that are at best inefficient and at worst perhaps fostering serious corruption. Eliminating such practices will probably be good for both business as well as economy and for the poor. Our results on starting costs, in particular, are certainly consistent with this notion. At the same time, policymakers must not lose sight of the fact that spurring the private sector is unlikely to be driven by the elimination or alleviation of "government failures." Spurring the private sector will also require the elimination or alleviation of market failures through appropriately designed public–private partnerships. The critical challenge for private sector development is probably getting this design correct. More research in uncovering the nature and processes that can lead to more effective public-private partnerships is likely to be a higher valueadded activity. It is also one that is likely to require in-depth, country-specific studies.

Finally, political freedom is not associated with either higher growth or lower poverty. Indeed, the evidence here seems to suggest that the delivery of good governance and regulations that facilitate the creation of new enterprises are more relevant for growth and poverty reduction than the nature of the political system per se.

# **SELECTED REFERENCES**

Acemoglu, D., J. Robinson, and S. Johnson. 2001. "The Colonial Origins of Comparative Development: An Empirical Investigation." *American Economic Review* 91(December):1369–401.

Asian Development Bank (ADB). 2004. Key Indicators 2004. Manila.

- ——. 2006. Poverty and Development Indicators Database. Available: http://lxapp1.asiandevbank.org:8030/ sdbs/povindex.jsp.
- ------. 2000. Private Sector Development Strategy: Promoting the Private Sector to Support Growth and Reduce Poverty. Manila.
- Barro, R. J. 1997. Determinants of Economic Growth. Cambridge, MA: MIT Press.
- Besley, T., and R. Burgess. 2003. "Can Labor Regulation Hinder Economic Performance? Evidence from India." London School of Economics. Processed.
- Bhalla, S. 2002. "Imagine There's No Country: Poverty, Inequality, and Growth in the Era of Globalization." Institute for International Economics, Washington, DC.
- Bhagwati, J. 2004. In Defense of Globalization. New York: Oxford University Press.

Chen, S., and M. Ravallion. 2004. "How Have the World's Poorest Fared Since the Early 1980s?" World Bank, Washington, DC. Available: www.worldbank.org/research/povmonitor/MartinPapers/ How\_have\_the\_poorest\_fared\_since\_the\_early\_1980s.pdf.

- Djankov, S., O. Hart, T. Nenova, and A. Shleifer. 2003. "Efficiency in Bankruptcy." Department of Economics, Harvard University, Massachusetts.
- Djankov, S., R. La Porta, F. Lopez-de-Silanes, and A. Shleifer. 2002. "The Regulation Entry." *Quarterly Journal* of Economics 117:1–37.

—. 2003. The Regulation of Labor. NBER Working Paper No. 9756, National Bureau of Economic Research, Cambridge, Massachusetts.

Dollar, D. 1992. "Outward-Oriented Developing Economies Really Do Grow More Rapidly: Evidence from 95 LDCs, 1976-1985." *Economic Development and Cultural Change* 40(3, April):523–44.

Dollar, D., and A. Kraay. 2001. *Growth is Good for the Poor*. Development Research Group, World Bank, Washington, DC.

<sup>——. 2003. &</sup>quot;Institutions, Trade and Growth." *Journal of Monetary Economics* 50(1):133–62.

- Hasan, R., M. G. Quibria, and Y. S. Kim. 2003. Poverty and Economic Freedom: Evidence from Cross-Country Data. Economics Study Area Working Paper No. 60, East-West Center, Honolulu.
- Easterly, W., and R. Levine. 2001. It's Not Factor Accumulation: Stylized Facts and Growth Models." *The World Bank Economic Review* 15(2):177–219.
- Edwards, S. 1998. "Openness, Productivity and Growth: What Do We Really Know." *The Economic Journal* 108(447):383-98.

Frankel, J., and D. Romer. 1999. "Does Trade Cause Growth?" American Economic Review 89(3, June):379-99.

- Freedom House. 2003. *Freedom In the World, Annual Survey of Political Rights and Civil Liberties*. Available: http://www.freedomhouse.org.
- Glaeser, E., R. La Porta, F. Lopez-de-Silanes, and A. Shleifer. 2004. Do Institutions Cause Growth. NBER Working Paper 10568, National Bureau of Economic Research, Cambridge, Massachusetts. Available: http:// www.nber.org/papers/w10568.
- Hall, R., and C. I. Jones. 1999. "Why Do Some Countries Produce So Much More Output Per Worker Than Others?" *The Quarterly Journal of Economics* 114(1, February):83–116.
- Hasan, R., M. G. Quibria, and Y. Kim 2002. "Absolute Poverty and the Role of Policy: Evidence from Cross-Country Data." East-West Center. Processed.
- Heritage Foundation. 2003. Annual Report. Washington, DC.
- International Labor Organization. 2006. LABORSTA Database. Available: http://www.ilo.org.
- Irwin, D., and M. J. Tervo. 2002. "Does Trade Raise Income? Evidence from the Twentieth Century." *Journal* of International Economics 58 (October):1–18.
- Jones, C. I. 1995. "Time Series Tests of Endogenous Growth Models." *Quarterly Journal of Economics* 110(May):495–525.
- Kaufman, D., A. Kraay, and P. Zaido-Lobaton. 1999. Governance Matters. World Bank Working Paper No. 2196, Washington, DC.
- ———. 2002. Governance Matters II: Updated Indicators for 2000/01. World Bank Working Paper No. 2772, Washington, DC.
- Keefer, P., and S. Knack. 1995. "Institutions and Economic Performance: Cross-Country Tests Using Alternative Institutional Measures." *Economics and Politics* 7(3):207–27.
- Mauro, P. 1995. "Corruption and Growth." Quarterly Journal of Economics 110:681-712.
- North, D. C. 1981. Structure and Change in Economic History. New York and London: Norton.
- Qian, Y. 2003. "How Reform worked in China." In D. Rodrik, ed., In Search of Prosperity: Analytic Narratives on Economic Growth. Princeton, NJ: Princeton University Press.
- Quibria, M. G. 2002. Growth and Poverty: Lessons from the East Asian Miracle Revisited. Research Paper Series No. 33, Asian Development Bank Institute, Tokyo.
- Ravallion, M. 2001. "Measuring Aggregate Welfare in Developing Countries: How Well Do National Accounts and Surveys Agree?" World Bank, Washington, DC. Processed.
- Ravallion, M., and G. Datt. 1999. When Is Growth Pro-poor? Evidence from the Diverse Experience of India's States. Policy Research Working Paper No. 2263, World Bank, Washington, DC.
- Rodrik, D. 1999. "Institutions for High-Quality Growth: What They Are and How to Acquire Them." Harvard University, Massachusetts. Processed.
- ———, 2003. "Growth Strategies." John F. Kennedy School of Government, Harvard University. Cambridge, MA.
- ——. 2004. "Getting Institutions Right." Harvard University, Massachusetts. Processed.

- Rodrik, D., and F. Rodriguez. 2001. "Trade Policy and Economic Growth: A Skeptic's Guide to the Cross-National Evidence." In B. Bernanke and K. S. Rogoff, eds., *Macroeconomics Annual 2000*. Cambridge: MIT Press for National Bureau of Economic Research.
- Rodrik, D., A. Subramanian, and F. Trebbi. 2002. "Institutions Rule: The Primacy of Institutions over Geography and Integration in Economic Development." Harvard University, Cambridge, MA. Available: http:// ksghome.harvard.edu/~drodrik/institutionsrule,%205.0.pdf.
- Sachs, J. D., and A. Warner. 1995. "Economic Reform and the Process of Global Integration." *Brookings Papers* on Economic Activity 1:1–118.
- Sala-I-Martin, X. 2002. The World Distribution of Income (Estimated from Individual Country Distributions). Economics Working Paper 615, Department of Economics and Business, Universitat Pompeu Fabra, Barcelona.
- Smith, A. 1976. An Inquiry into the Nature and Causes of the Wealth of Nations. E. Canaan, ed. Chicago: University of Chicago Press.
- Winters, L. A., N. McCulloch, and A. McKay. 2004. "Trade Liberalization and Poverty: The Evidence So Far." Journal of Economic Literature 42(1, March):72–115.
- Wacziarg, R., and K. H. Welch. 2003. Trade Liberalization and Growth: New Evidence. NBER Working Paper No. 10152, National Bureau of Economic Research, Massachusetts.
- World Bank. 2003. World Development Indicators. Washington, DC.
- ————. 2004a. Doing Business in 2005: Removing Obstacle to Growth. Washington, DC. Available: http:// rri.worldbank.org/doingbusiness.
- World Bank. 2004b. PovcalNet Database. Washington, DC. Available: http://iresearch.worldbank.org/PovcalNet/jsp/index.jsp.

# PUBLICATIONS FROM THE ECONOMICS AND RESEARCH DEPARTMENT

#### **ERD WORKING PAPER SERIES (WPS)**

(Published in-house; Available through ADB Office of External Relations; Free of Charge)

No.	1	Capitalizing on Globalization —Barry Eichengreen January 2002
No.	2	Policy-based Lending and Poverty Reduction:
		and Options
		-Richard Bolt and Manabu Fujimura, January
No.	3	The Automotive Supply Chain: Global Trends
		and Asian Perspectives
No.	4	-Francisco Veloso and Rajiv Kumar, January 2002 International Competitiveness of Asian Firms:
		An Analytical Framework
No.	5	The International Competitiveness of Asian
		Economies in the Apparel Commodity Chain
No.	6	Monetary and Financial Cooperation in East
	-	Asia—The Chiang Mai Initiative and Beyond
No	7	-Pradumna B. Rana, February 2002 Probing Beneath Cross-pational Averages: Poverty
140.	'	Inequality, and Growth in the Philippines
		—Arsenio M. Balisacan and Ernesto M. Pernia,
	0	March 2002
No.	8	-Anil B Declalibar April 2002
No.	9	Microfinance in Northeast Thailand: Who Benefits
		and How Much?
No	10	-Brett E. Coleman, April 2002
10.	10	Developing Asia
		-Anil B. Deolalikar, Alex B. Brilliantes, Jr., Rashav Gaiha, Ernesto M. Pernia, Mary Racelis
		with the assistance of Marita Concepcion Castro-
		Guevara, Liza L. Lim, Pilipinas F. Quising, May
No.	11	The European Social Model: Lessons for
1.00		Developing Countries
		-Assar Lindbeck, May 2002
No.	12	Costs and Benefits of a Common Currency for ASEAN
		—Srinivasa Madhur, May 2002
No.	13	Monetary Cooperation in East Asia: A Survey
No.	14	Toward A Political Economy Approach
		to Policy-based Lending
		—George Abonyi, May 2002
No.	15	A Framework for Establishing Priorities in a Country Poverty Reduction Strategy
		-Ron Duncan and Steve Pollard, June 2002
No.	16	The Role of Infrastructure in Land-use Dynamics
		and Rice Production in Viet Nam's Mekong River Delta
		-Christopher Edmonds, July 2002
No.	17	Effect of Decentralization Strategy on
		-Kanoknan Lao-Araya, August 2002
No.	18	Poverty and Patterns of Growth
		—Rana Hasan and M. G. Quibria, August 2002
No.	19	Why are Some Countries Richer than Others? A Reassessment of Mankiw-Romer-Weil's Test of

the Neoclassical Growth Model

- -Jesus Felipe and John McCombie, August 2002 No. 20 Modernization and Son Preference in People's Republic of China -Robin Burgess and Juzhong Zhuang, September 2002
- No. 21 The Doha Agenda and Development: A View from the Uruguay Round —J. Michael Finger, September 2002
- No. 22 Conceptual Issues in the Role of Education
   Decentralization in Promoting Effective Schooling in Asian Developing Countries
   —Jere R. Behrman, Anil B. Deolalikar, and Lee-Ying Son, September 2002
- No. 23 Promoting Effective Schooling through Education Decentralization in Bangladesh, Indonesia, and Philippines
   —Jere R. Behrman, Anil B. Deolalikar, and Lee-Ying Son, September 2002
- No. 24 Financial Opening under the WTO Agreement in Selected Asian Countries: Progress and Issues —Yun-Hwan Kim, September 2002
- No. 25 Revisiting Growth and Poverty Reduction in Indonesia: What Do Subnational Data Show? —Arsenio M. Balisacan, Ernesto M. Pernia, and Abuzar Asra, October 2002
- No. 26 Causes of the 1997 Asian Financial Crisis: What Can an Early Warning System Model Tell Us? —Juzhong Zhuang and J. Malcolm Dowling, October 2002
- No. 27 Digital Divide: Determinants and Policies with Special Reference to Asia —M. G. Quibria, Shamsun N. Ahmed, Ted Tschang, and Mari-Len Reyes-Macasaquit, October 2002
- No. 28 Regional Cooperation in Asia: Long-term Progress, Recent Retrogression, and the Way Forward —Ramgopal Agarwala and Brahm Prakash, October 2002
- No. 29 How can Cambodia, Lao PDR, Myanmar, and Viet Nam Cope with Revenue Lost Due to AFTA Tariff Reductions?
- —Kanokpan Lao-Araya, November 2002
   No. 30 Asian Regionalism and Its Effects on Trade in the 1980s and 1990s
   —Ramon Clarete, Christopher Edmonds, and Jessica Seddon Wallack, November 2002
- No. 31 New Economy and the Effects of Industrial Structures on International Equity Market Correlations --Cyn-Young Park and Jaejoon Woo, December 2002
- No. 32 Leading Indicators of Business Cycles in Malaysia and the Philippines —Wenda Zhang and Juzhong Zhuang, December 2002
- No. 33 Technological Spillovers from Foreign Direct Investment—A Survey —Emma Xiaoqin Fan, December 2002

No.	34	Economic Openness and Regional Development in the Philippines
No	35	-Ernesto M. Pernia and Pilipinas F. Quising, January 2003 Bond Market Development in East Asia:
1101		Issues and Challenges —Raul Fabella and Srinivasa Madhur, January
No.	36	2003 Environment Statistics in Central Asia: Progress and Prospects
		-Robert Ballance and Bishnu D. Pant, March 2003
No.	37	Electricity Demand in the People's Republic of China: Investment Requirement and Environmental Impact
No.	38	-Bo Q. Lin, March 2003 Foreign Direct Investment in Developing Asia: Trends, Effects, and Likely Issues for the Forthcoming WTO Negotiations
		-Douglas H. Brooks, Emma Xiaoqin Fan, and Lea R. Sumulong, April 2003
No.	39	The Political Economy of Good Governance for Poverty Alleviation Policies
No.	40	Narayan Lakshman, April 2003 The Puzzle of Social Capital A Critical Review
No.	41	-M. G. Quibria, May 2003 Industrial Structure, Technical Change, and the Role of Government in Development of the Electronics and Information Industry in
NT	40	Taipei,China —Yeo Lin, May 2003
No.	42	in Viet Nam —Arsenio M. Balisacan, Ernesto M. Pernia, and
No.	43	Gemma Esther B. Estrada, June 2003 Why Has Income Inequality in Thailand Increased? An Analysis Using 1975-1998 Surveys
No.	44	— <i>Taizo Motonishi, June 2003</i> Welfare Impacts of Electricity Generation Sector Reform in the Philippines
No.	45	Natsuko Toba, June 2003 A Review of Commitment Savings Products in Developing Countries Nava Ashrof Nathalia Cone, Dean S. Karlan
No.	46	and Wesley Yin, July 2003 Local Government Finance, Private Resources, and Local Credit Markets in Asia —Roberto de Vera and Yun-Hwan Kim, October
No.	47	2003 Excess Investment and Efficiency Loss During Reforms: The Case of Provincial-level Fixed-Asset Investment in People's Republic of China
No.	48	-Duo Qin and Haiyan Song, October 2003 Is Export-led Growth Passe? Implications for Developing Asia
No.	49	-Jesus Felipe, December 2003 Changing Bank Lending Behavior and Corporate Financing in Asia-Some Research Issues -Emma Xiaoqin Fan and Akiko Terada-Hagiwara
No.	50	Is People's Republic of China's Rising Services Sector Leading to Cost Disease?
No.	51	Poverty Estimates in India: Some Key Issues
No.	52	Restructuring and Regulatory Reform in the Power Sector: Review of Experience and Issues
No.	53	Peter Choynowski, May 2004 Competitiveness, Income Distribution, and Growth in the Philippines: What Does the Long-run Fuidonce Show?
		-Jesus Felipe and Grace C. Sipin, June 2004

No. 54	Practices of Poverty Measurement and Poverty
	Profile of Bangladesh
	—Faizuddin Ahmed, August 2004

- No. 55 Experience of Asian Asset Management Companies: Do They Increase Moral Hazard? —Evidence from Thailand —Akiko Terada-Hagiwara and Gloria Pasadilla, September 2004
- No. 56 Viet Nam: Foreign Direct Investment and Postcrisis Regional Integration —Vittorio Leproux and Douglas H. Brooks, September 2004
- No. 57 Practices of Poverty Measurement and Poverty Profile of Nepal
- -Devendra Chhetry, September 2004 No. 58 Monetary Poverty Estimates in Sri Lanka: Selected Issues -Neranjana Gunetilleke and Dinushka Senanayake, October 2004
- No. 59 Labor Market Distortions, Rural-Urban Inequality, and the Opening of People's Republic of China's Economy
- -Thomas Hertel and Fan Zhai, November 2004 No. 60 Measuring Competitiveness in the World's Smallest
- Economies: Introducing the SSMECI —Ganeshan Wignaraja and David Joiner, November 2004
- No. 61 Foreign Exchange Reserves, Exchange Rate Regimes, and Monetary Policy: Issues in Asia —Akiko Terada-Hagiwara, January 2005
- No. 62 A Small Macroeconometric Model of the Philippine Economy

   Geoffrey Ducanes, Marie Anne Cagas, Duo Qin, Pilipinas Quising, and Nedelyn Magtibay-Ramos, January 2005
- No. 63 Developing the Market for Local Currency Bonds by Foreign Issuers: Lessons from Asia —Tobias Hoschka, February 2005
- No. 64 Empirical Assessment of Sustainability and Feasibility of Government Debt: The Philippines Case
   —Duo Qin, Marie Anne Cagas, Geoffrey Ducanes, Nedelyn Magtibay-Ramos, and Pilipinas Quising,
- February 2005 No. 65 Poverty and Foreign Aid Evidence from Cross-Country Data —Abuzar Asra, Gemma Estrada, Yangseom Kim, and M. G. Quibria, March 2005
- No. 66 Measuring Efficiency of Macro Systems: An Application to Millennium Development Goal Attainment —Ajay Tandon, March 2005
- No. 67 Banks and Corporate Debt Market Development —Paul Dickie and Emma Xiaogin Fan, April 2005
- No. 68 Local Currency Financing—The Next Frontier for MDBs?
- -Tobias C. Hoschka, April 2005 No. 69 Export or Domestic-Led Growth in Asia?
- -Jesus Felipe and Joseph Lim, May 2005 No. 70 Policy Reform in Viet Nam and the Asian Development Bank's State-owned Enterprise Reform and Corporate Governance Program Loan -George Abonyi, August 2005
- No. 71 Policy Reform in Thailand and the Asian Development Bank's Agricultural Sector Program Loan —George Abonyi, September 2005
- No. 72 Can the Poor Benefit from the Doha Agenda? The Case of Indonesia —Douglas H. Brooks and Guntur Sugiyarto, October 2005
- No. 73 Impacts of the Doha Development Agenda on People's Republic of China: The Role of Complementary Education Reforms —Fan Zhai and Thomas Hertel, October 2005

No. 74	Growth and Trade Horizons for Asia: Long-term
	Forecasts for Regional Integration
	-David Roland-Holst, Jean-Pierre Verbiest, and
	Fan Zhai, November 2005
No. 75	Macroeconomic Impact of HIV/AIDS in the Asian
	and Pacific Region
	—Ajay Tandon, November 2005
No. 76	Policy Reform in Indonesia and the Asian
	Development Bank's Financial Sector Governance
	Reforms Program Loan
	—George Abonyi, December 2005
No. 77	Dynamics of Manufacturing Competitiveness in
	South Asia: ANalysis through Export Data
	—Hans-Peter Brunner and Massimiliano Calì,
	December 2005
No. 78	Trade Facilitation

—Teruo Ujiie, January 2006

#### **ERD POLICY BRIEF SERIES (PBS)**

(Published in-house; Available through ADB Office of External Relations; Free of charge)

No. 1	Is Growth Good Enough for the Poor?
	—Ernesto M. Pernia, October 2001
No. 2	India's Economic Reforms
	What Has Been Accomplished?
	What Remains to Be Done?
	—Arvind Panagariya, November 2001

- No. 3 Unequal Benefits of Growth in Viet Nam —Indu Bhushan, Erik Bloom, and Nguyen Minh Thang, January 2002
- No. 4 Is Volatility Built into Today's World Economy? —J. Malcolm Dowling and J.P. Verbiest, February 2002
- No. 5 What Else Besides Growth Matters to Poverty Reduction? Philippines —Arsenio M. Balisacan and Ernesto M. Pernia, February 2002
- No. 6 Achieving the Twin Objectives of Efficiency and Equity: Contracting Health Services in Cambodia —Indu Bhushan, Sheryl Keller, and Brad Schwartz, March 2002
- No. 7 Causes of the 1997 Asian Financial Crisis: What Can an Early Warning System Model Tell Us? —Juzhong Zhuang and Malcolm Dowling, June 2002
- No. 8 The Role of Preferential Trading Arrangements in Asia —Christopher Edmonds and Jean-Pierre Verbiest,
- No. 9 July 2002 No. 9 The Doha Round: A Development Perspective —Jean-Pierre Verbiest, Jeffrey Liang, and Lea Sumulong, July 2002
- No. 10 Is Economic Openness Good for Regional Development and Poverty Reduction? The Philippines —E. M. Pernia and Pilipinas Quising, October
- 2002 No. 11 Implications of a US Dollar Depreciation for Asian Developing Countries —Emma Fan, July 2002 No. 12 Dangers of Deflation —D. Brooks and Pilipinas Quising, December 2002 No. 13 Infrastructure and Poverty Reduction— What is the Connection? —Ifzal Ali and Ernesto Pernia, January 2003
- No. 14 Infrastructure and Poverty Reduction— Making Markets Work for the Poor —Xianbin Yao, May 2003
- No. 15 SARS: Economic Impacts and Implications —*Emma Xiaoqin Fan, May 2003*

- No. 79 An Assessment of Cross-country Fiscal Consolidation —Bruno Carrasco and Seung Mo Choi, February 2006
- No. 80 Central Asia: Mapping Future Prospects to 2015 —Malcolm Dowling and Ganeshan Wignaraja, April 2006
- No. 81 A Small Macroeconometric Model of the People's Republic of China —Duo Qin, Marie Anne Cagas, Geoffrey Ducanes, Nedelyn Magtibay-Ramos, Pilipinas Quising, Xin-Hua He, Rui Liu, and Shi-Guo Liu, June 2006
- No. 82 Institutions and Policies for Growth and Poverty Reduction: The Role of Private Sector Development —Rana Hasan, Devashish Mitra, and Mehmet Ulubasoglu, July 2006
- No. 16 Emerging Tax Issues: Implications of Globalization and Technology
- --Kanokpan Lao Araya, May 2003 No. 17 Pro-Poor Growth: What is It and Why is It
- Important? —Ernesto M. Pernia, May 2003
- No. 18 Public–Private Partnership for Competitiveness —Jesus Felipe, June 2003
- No. 19 Reviving Asian Economic Growth Requires Further Reforms —Ifzal Ali, June 2003
- No. 20 The Millennium Development Goals and Poverty: Are We Counting the World's Poor Right? —M. G. Quibria, July 2003
- No. 21 Trade and Poverty: What are the Connections? —Douglas H. Brooks, July 2003
- No. 22 Adapting Education to the Global Economy —Olivier Dupriez, September 2003
- No. 23 Avian Flu: An Economic Assessment for Selected Developing Countries in Asia —Jean-Pierre Verbiest and Charissa Castillo, March 2004
- No. 25 Purchasing Power Parities and the International Comparison Program in a Globalized World —Bishnu Pant, March 2004
- No. 26 A Note on Dual/Multiple Exchange Rates —*Emma Xiaoqin Fan, May 2004*
- No. 27 Inclusive Growth for Sustainable Poverty Reduction in Developing Asia: The Enabling Role of Infrastructure Development --Ifzal Ali and Xianbin Yao, May 2004
- No. 28 Higher Oil Prices: Asian Perspectives and Implications for 2004-2005 -Cyn-Young Park, June 2004
- No. 29 Accelerating Agriculture and Rural Development for Inclusive Growth: Policy Implications for Developing Asia —Richard Bolt, July 2004
- No. 30 Living with Higher Interest Rates: Is Asia Ready? —Cyn-Young Park, August 2004
- No. 31 Reserve Accumulation, Sterilization, and Policy Dilemma
- —Akiko Terada-Hagiwara, October 2004
   No. 32 The Primacy of Reforms in the Emergence of People's Republic of China and India
   —Ifzal Ali and Emma Xiaoqin Fan, November 2004
- No. 33 Population Health and Foreign Direct Investment: Does Poor Health Signal Poor Government Effectiveness? —Ajay Tandon, January 2005

No. 34	Financing Infrastructure Development: Asian	No. 40	Highlighting Poverty as Vulnerability: The 2005
	Developing Countries Need to Tap Bond Markets		Earthquake in Pakistan
	More Rigorously		-Rana Hasan and Ajay Tandon, October 2005
	—Yun-Hwan Kim, February 2005	No. 41	Conceptualizing and Measuring Poverty as Vulnerabil-
No. 35	Attaining Millennium Development Goals in		ity: Does It Make a Difference?
	Health: Isn't Economic Growth Enough?		–Ajay Tandon and Rana Hasan, October 2005
	-Ajay Tandon, March 2005	No. 42	Potential Economic Impact of an Avian Flu Pandemic on
No. 36	Instilling Credit Culture in State-owned Banks-		Asia
	Experience from Lao PDR		-Erik Bloom, Vincent de Wit, and Mary Jane Carangal-
	—Robert Boumphrey, Paul Dickie, and Samiuela		San Jose, November 2005
	Tukuafu, April 2005	No. 43	Creating Better and More Jobs in Indonesia: A
No. 37	Coping with Global Imbalances and Asian		Blueprint for Policy Action
	Currencies		—Guntur Sugiyarto, December 2005
	-Cyn-Young Park, May 2005	No. 44	The Challenge of Job Creation in Asia
No. 38	Asia's Long-term Growth and Integration:		—Jesus Felipe and Rana Hasan, April 2006
	Reaching beyond Trade Policy Barriers	No. 45	International Payments Imbalances
	-Douglas H. Brooks, David Roland-Holst, and Fan		—Jesus Felipe, Frank Harrigan, and Aashish Mehta,
	Zhai, September 2005		April 2006
No. 39	Competition Policy and Development	No. 46	Improving Primary Enrollment Rates among the Poor
	-Douglas H. Brooks, October 2005		-Ajay Tandon, July 2006

#### **ERD TECHNICAL NOTE SERIES (TNS)**

(Published in-house; Available through ADB Office of External Relations; Free of Charge)

No. 1	Contingency Calculations for Environmental
	Impacts with Unknown Monetary Values
	-David Dole, February 2002
No. 2	Integrating Risk into ADB's Economic Analysis
	of Projects
	—Nigel Rayner, Anneli Lagman-Martin,
	and Keith Ward, June 2002
No. 3	Measuring Willingness to Pay for Electricity
	-Peter Choynowski, July 2002
No. 4	Economic Issues in the Design and Analysis of a
	Wastewater Treatment Project
	—David Dole, July 2002
No. 5	An Analysis and Case Study of the Role of
	Environmental Economics at the Asian
	Development Bank
	—David Dole and Piya Abeygunawardena,
	September 2002
No. 6	Economic Analysis of Health Projects: A Case Study
	in Cambodia
	-Erik Bloom and Peter Choynowski, May 2003
No. 7	Strengthening the Economic Analysis of Natural
	Resource Management Projects
	—Keith Ward, September 2003
No. 8	Testing Savings Product Innovations Using an
	Experimental Methodology
	—Nava Ashraf, Dean S. Karlan, and Wesley Yin,
	November 2003
No. 9	Setting User Charges for Public Services: Policies
	and Practice at the Asian Development Bank
	—David Dole, December 2003

- No. 10 Beyond Cost Recovery: Setting User Charges for Financial, Economic, and Social Goals —David Dole and Ian Bartlett, January 2004
- No. 11 Shadow Exchange Rates for Project Economic Analysis: Toward Improving Practice at the Asian Development Bank —Anneli Lagman-Martin, February 2004
- No. 12 Improving the Relevance and Feasibility of Agriculture and Rural Development Operational Designs: How Economic Analyses Can Help —Richard Bolt, September 2005
- No. 13 Assessing the Use of Project Distribution and Poverty Impact Analyses at the Asian Development Bank —Franklin D. De Guzman, October 2005
- No. 14 Assessing Aid for a Sector Development Plan: Economic Analysis of a Sector Loan —David Dole, November 2005
- No. 15 Debt Management Analysis of Nepal's Public Debt —Sungsup Ra, Changyong Rhee, and Joon-Ho Hahm, December 2005
- No. 16 Evaluating Microfinance Program Innovation with Randomized Control Trials: An Example from Group Versus Individual Lending —Xavier Giné, Tomoko Harigaya,Dean Karlan, and Binh T. Nguyen, March 2006
- No. 17 Setting User Charges for Urban Water Supply: A Case Study of the Metropolitan Cebu Water District in the Philippines —David Dole and Edna Balucan, June 2006

#### SPECIAL STUDIES, COMPLIMENTARY

#### (Available through ADB Office of External Relations)

- 1. Improving Domestic Resource Mobilization Through Financial Development: Overview September 1985
- 2. Improving Domestic Resource Mobilization Through Financial Development: Bangladesh July 1986
- 3. Improving Domestic Resource Mobilization Through Financial Development: Sri Lanka April 1987
- 4. Improving Domestic Resource Mobilization Through Financial Development: India December 1987
- Financing Public Sector Development Expenditure in Selected Countries: Overview January 1988
   Study of Selected Industries: A Brief Report
- April 1988
- 7. Financing Public Sector Development Expenditure in Selected Countries: Bangladesh *June 1988*
- 8. Financing Public Sector Development Expenditure in Selected Countries: India June 1988
- 9. Financing Public Sector Development Expenditure in Selected Countries: Indonesia June 1988
- 10. Financing Public Sector Development Expenditure in Selected Countries: Nepal June 1988
- 11. Financing Public Sector Development Expenditure in Selected Countries: Pakistan June 1988
- 12. Financing Public Sector Development Expenditure in Selected Countries: Philippines June 1988
- 13. Financing Public Sector Development Expenditure in Selected Countries: Thailand June 1988
- Towards Regional Cooperation in South Asia: ADB/EWC Symposium on Regional Cooperation in South Asia February 1988
- 15. Evaluating Rice Market Intervention Policies: Some Asian Examples April 1988
- 16. Improving Domestic Resource Mobilization Through Financial Development: Nepal November 1988
- 17. Foreign Trade Barriers and Export Growth September 1988
- 18. The Role of Small and Medium-Scale Industries in the Industrial Development of the Philippines April 1989

- The Role of Small and Medium-Scale Manufacturing Industries in Industrial Development: The Experience of Selected Asian Countries January 1990
- 20. National Accounts of Vanuatu, 1983-1987 January 1990
- 21. National Accounts of Western Samoa, 1984-1986 February 1990
- 22. Human Resource Policy and Economic Development: Selected Country Studies July 1990
- 23. Export Finance: Some Asian Examples September 1990
- 24. National Accounts of the Cook Islands, 1982-1986 September 1990
- 25. Framework for the Economic and Financial Appraisal of Urban Development Sector Projects January 1994
- 26. Framework and Criteria for the Appraisal and Socioeconomic Justification of Education Projects January 1994
- 27. Investing in Asia 1997 (Co-published with OECD)
- 28. The Future of Asia in the World Economy 1998 (Copublished with OECD)
- Financial Liberalisation in Asia: Analysis and Prospects 1999 (Co-published with OECD)
- Sustainable Recovery in Asia: Mobilizing Resources for Development 2000 (Co-published with OECD)
- 31. Technology and Poverty Reduction in Asia and the Pacific 2001 (Co-published with OECD)
- 32. Asia and Europe 2002 (Co-published with OECD)
- 33. Economic Analysis: Retrospective 2003
- 34. Economic Analysis: Retrospective: 2003 Update 2004
- 35. Development Indicators Reference Manual: Concepts and Definitions 2004
- 35. Investment Climate and Productivity Studies Philippines: Moving Toward a Better Investment Climate 2005

The Road to Recovery: Improving the Investment Climate in Indonesia  $2005\,$ 

Sri Lanka: Improving the Rural and Urban Investment Climate 2005

#### **OLD MONOGRAPH SERIES**

(Available through ADB Office of External Relations; Free of charge)

#### EDRC REPORT SERIES (ER)

No. 1	ASEAN and the Asian Development Bank
	—Seiji Naya, April 1982
No. 2	Development Issues for the Developing East
	and Southeast Asian Countries
	and International Cooperation
	—Seiji Naya and Graham Abbott, April 1982
No. 3	Aid, Savings, and Growth in the Asian Region
	-J. Malcolm Dowling and Ulrich Hiemenz,
	April 1982
No. 4	Development-oriented Foreign Investment
	and the Role of ADB
	—Kiyoshi Kojima, April 1982
No. 5	The Multilateral Development Banks
	and the International Economy's Missing
	Public Sector
	—John Lewis, June 1982
No. 6	Notes on External Debt of DMCs
	—Evelyn Go, July 1982
No. 7	Grant Element in Bank Loans
	—Dal Hyun Kim, July 1982
No. 8	Shadow Exchange Rates and Standard
	<b>Conversion Factors in Project Evaluation</b>

 —Peter Warr, September 1982
 No. 9 Small and Medium-Scale Manufacturing Establishments in ASEAN Countries: Perspectives and Policy Issues
 —Mathias Bruch and Ulrich Hiemenz, January 1983

- No. 10 A Note on the Third Ministerial Meeting of GATT —Jungsoo Lee, January 1983
- No. 11 Macroeconomic Forecasts for the Republic of China, Hong Kong, and Republic of Korea —J.M. Dowling, January 1983
- No. 12 ASEAN: Economic Situation and Prospects —Seiji Naya, March 1983
- No. 13 The Future Prospects for the Developing Countries of Asia —Seiji Naya, March 1983
- No. 14 Energy and Structural Change in the Asia-Pacific Region, Summary of the Thirteenth Pacific Trade and Development Conference —Seiji Naya, March 1983
- No. 15 A Survey of Empirical Studies on Demand for Electricity with Special Emphasis on Price

Elasticity of Demand —Wisarn Pupphavesa, June 1983 Determinants of Paddy Production in Indonesia: No. 16 1972-1981-A Simultaneous Equation Model Approach —T.K. Jayaraman, June 1983 No. 17 The Philippine Economy: Economic Forecasts for 1983 and 1984 -J.M. Dowling, E. Go, and C.N. Castillo, June 1983 No. 18 Economic Forecast for Indonesia -J.M. Dowling, H.Y. Kim, Y.K. Wang, and C.N. Castillo, June 1983 No. 19 Relative External Debt Situation of Asian Developing Countries: An Application of Ranking Method -Jungsoo Lee, June 1983 New Evidence on Yields, Fertilizer Application, No. 20 and Prices in Asian Rice Production —William James and Teresita Ramirez, July 1983 No. 21 Inflationary Effects of Exchange Rate Changes in Nine Asian LDCs -Pradumna B. Rana and J. Malcolm Dowling, Jr., December 1983 Effects of External Shocks on the Balance No. 22 of Payments, Policy Responses, and Debt Problems of Asian Developing Countries —Seiji Naya, December 1983 Changing Trade Patterns and Policy Issues: No. 23 The Prospects for East and Southeast Asian **Developing Countries** -Seiji Naya and Ulrich Hiemenz, February 1984 No. 24 Small-Scale Industries in Asian Economic Development: Problems and Prospects —Seiji Naya, February 1984 No. 25 A Study on the External Debt Indicators Applying Logit Analysis -Jungsoo Lee and Clarita Barretto, February 1984 Alternatives to Institutional Credit Programs No. 26 in the Agricultural Sector of Low-Income Countries –Jennifer Sour. March 1984 Economic Scene in Asia and Its Special Features No. 27 -Kedar N. Kohli, November 1984 No. 28 The Effect of Terms of Trade Changes on the Balance of Payments and Real National Income of Asian Developing Countries -Jungsoo Lee and Lutgarda Labios, January 1985 No. 29 Cause and Effect in the World Sugar Market: Some Empirical Findings 1951-1982 —Yoshihiro Iwasaki, February 1985 Sources of Balance of Payments Problem No. 30 in the 1970s: The Asian Experience -Pradumna Rana, February 1985 India's Manufactured Exports: An Analysis No. 31 of Supply Sectors —Ifzal Ali, February 1985 No. 32 Meeting Basic Human Needs in Asian **Developing Countries** -Jungsoo Lee and Emma Banaria, March 1985 No. 33 The Impact of Foreign Capital Inflow on Investment and Economic Growth in Developing Asia -Evelyn Go, May 1985 The Climate for Energy Development No. 34 in the Pacific and Asian Region: Priorities and Perspectives –V.V. Desai, April 1986 Impact of Appreciation of the Yen on No. 35 Developing Member Countries of the Bank -Jungsoo Lee, Pradumna Rana, and Ifzal Ali, Mav 1986 No. 36 Smuggling and Domestic Economic Policies in Developing Countries

-A.H.M.N. Chowdhury, October 1986 No. 37 Public Investment Criteria: Economic Internal Rate of Return and Equalizing Discount Rate —Ifzal Ali, November 1986 No. 38 Review of the Theory of Neoclassical Political Economy: An Application to Trade Policies —M.G. Quibria, December 1986 No. 39 Factors Influencing the Choice of Location: Local and Foreign Firms in the Philippines -E.M. Pernia and A.N. Herrin, February 1987 No. 40 A Demographic Perspective on Developing Asia and Its Relevance to the Bank —E.M. Pernia, May 1987 No. 41 Emerging Issues in Asia and Social Cost Benefit Analysis —I. Ali, September 1988 Shifting Revealed Comparative Advantage: No. 42 Experiences of Asian and Pacific Developing Countries -P.B. Rana, November 1988 No. 43 Agricultural Price Policy in Asia: Issues and Areas of Reforms -I. Ali. November 1988 Service Trade and Asian Developing Economies No. 44 -M.G. Quibria. October 1989 No. 45 A Review of the Economic Analysis of Power Projects in Asia and Identification of Areas of Improvement —I. Ali, November 1989 Growth Perspective and Challenges for Asia: No. 46 Areas for Policy Review and Research -I. Ali, November 1989 No. 47 An Approach to Estimating the Poverty Alleviation Impact of an Agricultural Project —I. Ali, January 1990 No. 48 Economic Growth Performance of Indonesia, the Philippines, and Thailand: The Human Resource Dimension -E.M. Pernia, January 1990 Foreign Exchange and Fiscal Impact of a Project: No. 49 A Methodological Framework for Estimation -I. Ali. February 1990 Public Investment Criteria: Financial No. 50

- No. 50 Public Investment Criteria: Financial and Economic Internal Rates of Return —I. Ali, April 1990
- No. 51 Evaluation of Water Supply Projects: An Economic Framework —Arlene M. Tadle, June 1990
- No. 52 Interrelationship Between Shadow Prices, Project Investment, and Policy Reforms: An Analytical Framework —I. Ali, November 1990
- No. 53 Issues in Assessing the Impact of Project and Sector Adjustment Lending —I. Ali, December 1990
- No. 54 Some Aspects of Urbanization and the Environment in Southeast Asia —Ernesto M. Pernia, January 1991
- No. 55 Financial Sector and Economic Development: A Survey —Jungsoo Lee, September 1991
- No. 56 A Framework for Justifying Bank-Assisted Education Projects in Asia: A Review of the Socioeconomic Analysis and Identification of Areas of Improvement —Etienne Van De Walle, February 1992
- No. 57 Medium-term Growth-Stabilization Relationship in Asian Developing Countries and Some Policy Considerations —Yun-Hwan Kim, February 1993
- No. 58 Urbanization, Population Distribution, and Economic Development in Asia —Ernesto M. Pernia, February 1993
- No. 59 The Need for Fiscal Consolidation in Nepal:

The Results of a Simulation

—Filippo di Mauro and Ronald Antonio Butiong, July 1993

- No. 60 A Computable General Equilibrium Model of Nepal —Timothy Buehrer and Filippo di Mauro, October 1993
   No. 61 The Role of Government in Export Expansion
- in the Republic of Korea: A Revisit —Yun-Hwan Kim, February 1994 No. 62 Rural Reforms, Structural Change, and Agricultural Growth in the People's Republic of China

-Bo Lin, August 1994

- No. 63 Incentives and Regulation for Pollution Abatement with an Application to Waste Water Treatment -Sudipto Mundle, U. Shankar, and Shekhar Mehta, October 1995
- No. 64 Saving Transitions in Southeast Asia —Frank Harrigan, February 1996
- No. 65 Total Factor Productivity Growth in East Asia: A Critical Survey —Jesus Felipe, September 1997
- No. 66 Foreign Direct Investment in Pakistan:
   Policy Issues and Operational Implications
   —Ashfaque H. Khan and Yun-Hwan Kim, July 1999
- No. 67 Fiscal Policy, Income Distribution and Growth —Sailesh K. Jha, November 1999

#### **ECONOMIC STAFF PAPERS (ES)**

No.	1	International Reserves: Factors Determining Needs and Adequacy
No.	2	Domestic Savings in Selected Developing Asian Countries
		—Basil Moore, assisted by A.H.M. Nuruddin Choudhury Sentember 1981
No.	3	Changes in Consumption. Imports and Exports
		of Oil Since 1973: A Preliminary Survey of
		the Developing Member Countries
		of the Asian Development Bank
		—Dal Hyun Kim and Graham Abbott, September
NT		
No.	4	By-Passed Areas, Regional Inequalities,
		Southoast Asian Countries
		-William James October 1981
No.	5	Asian Agriculture and Economic Development
		-William James, March 1982
No.	6	Inflation in Developing Member Countries:
		An Analysis of Recent Trends
		-A.H.M. Nuruddin Chowdhury and J. Malcolm
		Dowling, March 1982
No.	7	Industrial Growth and Employment in
		Developing Asian Countries: Issues and
		Illrich Hiemana March 1082
No	8	Petrodollar Recycling 1973-1980
110.	0	Part 1: Regional Adjustments and
		the World Economy
		—Burnham Campbell, April 1982
No.	9	Developing Asia: The Importance
		of Domestic Policies
		—Economics Office Staff under the direction of Seiji
NT	10	Naya, May 1982
INO.	10	Financial Development and Household
		Mobilization in Asian Developing Countries
		-Wan-Soon Kim July 1982
No.	11	Industrial Development: Role of Specialized
		Financial Institutions
		—Kedar N. Kohli, August 1982
No.	12	Petrodollar Recycling 1973-1980.
		Part II: Debt Problems and an Evaluation
		of Suggested Remedies
NT.	10	-Burnham Campbell, September 1982
INO.	13	Policy in Developing Countries
		-William James Sentember 1989
No	14	Small and Medium-Scale Manufacturing
		Establishments in ASEAN Countries:
		Perspectives and Policy Issues
		-Mathias Bruch and Ulrich Hiemenz, March 1983
No.	15	Income Distribution and Economic

		Growth in Developing Asian Countries —J. Malcolm Dowling and David Soo, March 1983
No.	16	Long-Run Debt-Servicing Capacity of Asian Developing Countries: An Application of Critical Interest Rate Approach
No.	17	External Shocks, Energy Policy, and Macroeconomic Performance of Asian
		-William James, July 1983
No.	18	The Impact of the Current Exchange Rate System on Trade and Inflation of Selected Developing Member Countries
No.	19	—Pradumna Rana, September 1983 Asian Agriculture in Transition: Key Policy Issues —William James, September 1983
No.	20	The Transition to an Industrial Economy
		-Harry T. Oshima, October 1983
No.	21	The Significance of Off-Farm Employment and Incomes in Post-War East Asian Growth —Harry T. Oshima, January 1984
No.	22	Income Distribution and Poverty in Selected
		Asian Countries —John Malcolm Dowling, Jr., November 1984
No.	23	ASEAN Economies and ASEAN Economic Cooperation
No.	24	-Narongchai Akrasanee, November 1984 Economic Analysis of Power Projects
No	95	-Nitin Desai, January 1985 Exports and Economia Crowth in the Asian Region
110.	20	-Pradumna Rana, February 1985
No.	26	Patterns of External Financing of DMCs —E. Go, May 1985
No.	27	Industrial Technology Development the Republic of Korea
		-S.Y. Lo, July 1985
No.	28	Risk Analysis and Project Selection: A Review of Practical Issues
No.	29	-J.K. Johnson, August 1985 Rice in Indonesia: Price Policy and Comparative
		Advantage
No.	30	-1. Ali, January 1986 Effects of Foreign Capital Inflows
		on Developing Countries of Asia —Jungsoo Lee, Pradumna B. Rana, and Yoshihiro
No.	31	Iwasaki, April 1986 Economic Analysis of the Environmental
		-John A. Dixon et al., EAPI, East-West Center,
No.	32	Science and Technology for Development:
		Role of the Bank —Kedar N. Kohli and Ifzal Ali, November 1986

No.	33	Satellite Remote Sensing in the Asian and Pacific Region
		—Mohan Sundara Rajan, December 1986
No.	34	Changes in the Export Patterns of Asian and
		Pacific Developing Countries: An Empirical
		Overview
		—Pradumna B. Rana, January 1987
No.	35	Agricultural Price Policy in Nepal
		-Gerald C. Nelson, March 1987
No.	36	Implications of Falling Primary Commodity
		Prices for Agricultural Strategy in the Philippines
		—Ifzal Ali, September 1987
No.	37	Determining Irrigation Charges: A Framework
		-Prabhakar B. Ghate, October 1987
No.	38	The Role of Fertilizer Subsidies in Agricultural
		Production: A Review of Select Issues
		-M.G. Quibria, October 1987
No.	39	Domestic Adjustment to External Shocks
		in Developing Asia
		—Jungsoo Lee, October 1987
No.	40	Improving Domestic Resource Mobilization
		through Financial Development: Indonesia
		—Philip Erquiaga, November 1987
No.	41	Recent Trends and Issues on Foreign Direct
		Investment in Asian and Pacific Developing
		Countries
		-P.B. Rana, March 1988
No.	42	Manufactured Exports from the Philippines:
		A Sector Profile and an Agenda for Reform
		—I. Ali, September 1988
No.	43	A Framework for Evaluating the Economic
		Benefits of Power Projects
		—I. Ali, August 1989
No.	44	Promotion of Manufactured Exports in Pakistan
		—Jungsoo Lee and Yoshihiro Iwasaki, September
		1989
No.	45	Education and Labor Markets in Indonesia:
		A Sector Survey
		—Ernesto M. Pernia and David N. Wilson,
		September 1989
No.	46	Industrial Technology Capabilities
		and Policies in Selected ADCs
		—Hiroshi Kakazu, June 1990
No.	47	Designing Strategies and Policies
		for Managing Structural Change in Asia
		—Ifzal Ali, June 1990

- No. 48 The Completion of the Single European Community Market in 1992: A Tentative Assessment of its Impact on Asian Developing Countries —J.P. Verbiest and Min Tang, June 1991
- No. 49 Economic Analysis of Investment in Power Systems —Ifzal Ali, June 1991
- No. 50 External Finance and the Role of Multilateral Financial Institutions in South Asia: Changing Patterns, Prospects, and Challenges —Jungsoo Lee, November 1991
- No. 51 The Gender and Poverty Nexus: Issues and Policies
- -M.G. Quibria, November 1993 No. 52 The Role of the State in Economic Development: Theory, the East Asian Experience, and the Malaysian Case -Jason Brown, December 1993
- No. 53 The Economic Benefits of Potable Water Supply Projects to Households in Developing Countries —Dale Whittington and Venkateswarlu Swarna, January 1994
- No. 54 Growth Triangles: Conceptual Issues and Operational Problems —Min Tang and Myo Thant, February 1994
- No. 55 The Emerging Global Trading Environment and Developing Asia
   —Arvind Panagariya, M.G. Quibria, and Narhari Rao, July 1996
- No. 56 Aspects of Urban Water and Sanitation in the Context of Rapid Urbanization in Developing Asia —Ernesto M. Pernia and Stella LF. Alabastro, September 1997
- No. 57 Challenges for Asia's Trade and Environment -Douglas H. Brooks, January 1998
- No. 58 Economic Analysis of Health Sector Projects-A Review of Issues, Methods, and Approaches —Ramesh Adhikari, Paul Gertler, and Anneli Lagman, March 1999
- No. 59 The Asian Crisis: An Alternate View —Rajiv Kumar and Bibek Debroy, July 1999
- No. 60 Social Consequences of the Financial Crisis in Asia —James C. Knowles, Ernesto M. Pernia, and Mary Racelis, November 1999

#### **OCCASIONAL PAPERS (OP)**

No. 1	Poverty in the People's Republic of China:
	Recent Developments and Scope
	for Bank Assistance
	—K.H. Moinuddin, November 1992
No. 2	The Eastern Islands of Indonesia: An Overview
	of Development Needs and Potential
	—Brien K. Parkinson, January 1993
No. 3	Rural Institutional Finance in Bangladesh
	and Nepal: Review and Agenda for Reforms
	—A.H.M.N. Chowdhury and Marcelia C. Garcia,
	November 1993
No. 4	Fiscal Deficits and Current Account Imbalances
	of the South Pacific Countries:
	A Case Study of Vanuatu
	—T.K. Jayaraman, December 1993
No. 5	Reforms in the Transitional Economies of Asia
	—Pradumna B. Rana, December 1993
No. 6	Environmental Challenges in the People's Republic
	of China and Scope for Bank Assistance
	—Elisabetta Capannelli and Omkar L. Shrestha,
	December 1993
No. 7	Sustainable Development Environment

and Poverty Nexus

	—K.F. Jalal, December 1993
No. 8	Intermediate Services and Economic
	Development: The Malaysian Example
	-Sutanu Behuria and Rahul Khullar, May 1994
No. 9	Interest Rate Deregulation: A Brief Survey
	of the Policy Issues and the Asian Experience
	-Carlos J. Glower, July 1994
No. 10	Some Aspects of Land Administration
	in Indonesia: Implications for Bank Operations
	—Sutanu Behuria, July 1994
No. 11	Demographic and Socioeconomic Determinants
	of Contraceptive Use among Urban Women in
	the Melanesian Countries in the South Pacific:
	A Case Study of Port Vila Town in Vanuatu
	—T.K. Jayaraman, February 1995
No. 12	Managing Development through
	Institution Building
	— Hilton L. Root, October 1995
No. 13	Growth, Structural Change, and Optimal
	Poverty Interventions
	—Shiladitya Chatterjee, November 1995

No. 14 Private Investment and Macroeconomic Environment in the South Pacific Island Countries: A Cross-Country Analysis

	—1.K. Jayaraman, October 1996
No. 15	The Rural-Urban Transition in Viet Nam:
	Some Selected Issues
	—Sudipto Mundle and Brian Van Arkadie, October
	1997
No. 16	A New Approach to Setting the Future
	Transport Agenda
	-Roger Allport, Geoff Key, and Charles Melhuish,
	June 1998
No. 17	Adjustment and Distribution:
	The Indian Experience
	-Sudipto Mundle and V.B. Tulasidhar, June 1998

No. 18 Tax Reforms in Viet Nam: A Selective Analysis —Sudipto Mundle, December 1998

- No. 19 Surges and Volatility of Private Capital Flows to Asian Developing Countries: Implications for Multilateral Development Banks —Pradumna B. Rana, December 1998
- No. 20 The Millennium Round and the Asian Economies: An Introduction —Dilip K. Das, October 1999
- No. 21 Occupational Segregation and the Gender Earnings Gap —Joseph E. Zveglich, Jr. and Yana van der Meulen Rodgers, December 1999
- No. 22 Information Technology: Next Locomotive of Growth? —Dilip K. Das, June 2000

Survey of the External Debt Situation

#### STATISTICAL REPORT SERIES (SR)

No. 9

No. 1	Estimates of the Total External Debt of
	the Developing Member Countries of ADB:
	1981-1983
	—I.P. David, September 1984
No. 2	Multivariate Statistical and Graphical
	Classification Techniques Applied
	to the Problem of Grouping Countries
	-I.P. David and D.S. Maligalig, March 1985
No. 3	Gross National Product (GNP) Measurement
	Issues in South Pacific Developing Member
	Countries of ADB
	—S.G. Tiwari, September 1985
No. 4	Estimates of Comparable Savings in Selected
	DMCs
	—Hananto Sigit, December 1985
No. 5	Keeping Sample Survey Design
	and Analysis Simple
	—I.P. David, December 1985
No. 6	External Debt Situation in Asian
	Developing Countries
	—I.P. David and Jungsoo Lee, March 1986
No. 7	Study of GNP Measurement Issues in the
	South Pacific Developing Member Countries.
	Part I: Existing National Accounts
	of SPDMCs–Analysis of Methodology
	and Application of SNA Concepts
	-P. Hodgkinson, October 1986
No. 8	Study of GNP Measurement Issues in the South
	Pacific Developing Member Countries.
	Part II: Factors Affecting Intercountry
	Comparability of Per Capita GNP
	-P. Hodgkinson, October 1986

	in Asian Developing Countries, 1985
NT 10	-Jungsoo Lee and I.P. David, April 1987
No. 10	A Survey of the External Debt Situation
	in Asian Developing Countries, 1986
	—Jungsoo Lee and I.P. David, April 1988
No. 11	Changing Pattern of Financial Flows to Asian
	and Pacific Developing Countries
	—Jungsoo Lee and I.P. David, March 1989
No. 12	The State of Agricultural Statistics in
	Southeast Asia
	—I.P. David, March 1989
No. 13	A Survey of the External Debt Situation
	in Asian and Pacific Developing Countries:
	1987-1988
	—Jungsoo Lee and I.P. David, July 1989
No. 14	A Survey of the External Debt Situation in
	Asian and Pacific Developing Countries: 1988-1989
	—Jungsoo Lee, May 1990
No. 15	A Survey of the External Debt Situation
	in Asian and Pacific Developing Countries: 1989-
	1992
	-Min Tang, June 1991
No. 16	Recent Trends and Prospects of External Debt
	Situation and Financial Flows to Asian
	and Pacific Developing Countries
	-Min Tang and Aludia Pardo, June 1992
No. 17	Purchasing Power Parity in Asian Developing
	Countries: A Co-Integration Test

 —Min Tang and Ronald Q. Butiong, April 1994
 No. 18 Capital Flows to Asian and Pacific Developing Countries: Recent Trends and Future Prospects
 —Min Tang and James Villafuerte, October 1995

#### SERIALS

(Available commercially through ADB Office of External Relations)

- Asian Development Outlook (ADO; annual) \$36.00 (paperback)
- Key Indicators of Developing Asian and Pacific Countries (KI; annual) \$35.00 (paperback)
- 3. Asian Development Review (ADR; semiannual) \$5.00 per issue; \$10.00 per year (2 issues)

#### SPECIAL STUDIES, CO-PUBLISHED

(Available commercially through Oxford University Press Offices, Edward Elgar Publishing, and Palgrave MacMillan)

#### FROM OXFORD UNIVERSITY PRESS:

Oxford University Press (China) Ltd 18th Floor, Warwick House East Taikoo Place, 979 King's Road Quarry Bay, Hong Kong Tel (852) 2516 3222 Fax (852) 2565 8491 E-mail: webmaster@oupchina.com.hk Web: www.oupchina.com.hk

- Informal Finance: Some Findings from Asia Prabhu Ghate et. al., 1992 \$15.00 (paperback)
- Mongolia: A Centrally Planned Economy in Transition Asian Development Bank, 1992 \$15.00 (paperback)
- Rural Poverty in Asia, Priority Issues and Policy Options Edited by M.G. Quibria, 1994
- \$25.00 (paperback)Growth Triangles in Asia: A New Approach
- Growth Thangles in Asia. A New Approach to Regional Economic Cooperation Edited by Myo Thant, Min Tang, and Hiroshi Kakazu Ist ed., 1994 \$36.00 (hardbound) Revised ed., 1998 \$55.00 (hardbound)
- Urban Poverty in Asia: A Survey of Critical Issues Edited by Ernesto Pernia, 1994 \$18.00 (paperback)
- Critical Issues in Asian Development: Theories, Experiences, and Policies Edited by M.G. Quibria, 1995 \$15.00 (paperback) \$36.00 (hardbound)
- Financial Sector Development in Asia Edited by Shahid N. Zahid, 1995 \$50.00 (hardbound)
   Financial Sector Development in Asia: Country Studies
- Financial Sector Development in Asia: Country Studies Edited by Shahid N. Zahid, 1995 \$55.00 (hardbound)
- Fiscal Management and Economic Reform in the People's Republic of China Christine P.W. Wong, Christopher Heady, and Wing T. Woo, 1995 \$15.00 (paperback)
- An Asian Perspective Edited by M.G. Quibria and J. Malcolm Dowling, 1996 \$50.00 (hardbound)
  12. The Bangladesh Economy in Transition Edited by M.G. Quibria, 1997
- \$20.00 (hardbound)13. The Global Trading System and Developing Asia
- Edited by Arvind Panagariya, M.G. Quibria, and Narhari Rao, 1997 \$55.00 (hardbound)
- 14. Social Sector Issues in Transitional Economies of Asia Edited by Douglas H. Brooks and Myo Thant, 1998

\$25.00 (paperback) \$55.00 (hardbound)

- Intergovernmental Fiscal Transfers in Asia: Current Practice and Challenges for the Future Edited by Yun-Hwan Kim and Paul Smoke, 2003 \$15.00 (paperback)
- Local Government Finance and Bond Markets *Edited by Yun-Hwan Kim, 2003* \$15.00 (paperback)

#### FROM EDWARD ELGAR:

Marston Book Services Limited PO Box 269, Abingdon Oxon OX14 4YN, United Kingdom Tel +44 1235 465500 Fax +44 1235 465555 Email: direct.order@marston.co.uk Web: www.marston.co.uk

 Reducing Poverty in Asia: Emerging Issues in Growth, Targeting, and Measurement Edited by Christopher M. Edmonds, 2003

#### FROM PALGRAVE MACMILLAN:

Palgrave Macmillan Ltd Houndmills, Basingstoke Hampshire RG21 6XS, United Kingdom Tel: +44 (0)1256 329242 Fax: +44 (0)1256 479476 Email: orders@palgrave.com Web: www.palgrave.com/home/

- 1. Labor Markets in Asia: Issues and Perspectives Edited by Jesus Felipe and Rana Hasan, 2006
- 2. Competition Policy and Development in Asia Edited by Douglas H. Brooks and Simon Evenett, 2005
- Managing FDI in a Globalizing Economy Asian Experiences
- Edited by Douglas H. Brooks and Hal Hill, 2004
- Poverty, Growth, and Institutions in Developing Asia Edited by Ernesto M. Pernia and Anil B. Deolalikar, 2003

#### SPECIAL STUDIES, IN-HOUSE

#### (Available commercially through ADB Office of External Relations)

- Rural Poverty in Developing Asia Edited by M.G. Quibria Vol. 1: Bangladesh, India, and Sri Lanka, 1994 \$35.00 (paperback) Vol. 2: Indonesia, Republic of Korea, Philippines, and Thailand, 1996 \$35.00 (paperback)
   Gender Indicators of Developing Asian and Pacific
- Gender Indicators of Developing Asian and Pacific Countries Asian Development Bank, 1993 \$25.00 (paperback)
- External Shocks and Policy Adjustments: Lessons from the Gulf Crisis Edited by Naved Hamid and Shahid N. Zahid, 1995
- \$15.00 (paperback)
  4. Indonesia-Malaysia-Thailand Growth Triangle: Theory to Practice
- Edited by Myo Thant and Min Tang, 1996 \$15.00 (paperback) 5. Emerging Asia: Changes and Challenges
- 5. Emerging Asia: Changes and Chanlenges Asian Development Bank, 1997 \$30.00 (paperback)
- Asian Exports Edited by Dilip Das, 1999 \$35.00 (paperback) \$55.00 (hardbound)
- Development of Environment Statistics in Developing Asian and Pacific Countries Asian Development Bank, 1999 \$30.00 (paperback)
- Mortgage-Backed Securities Markets in Asia Edited by S.Ghon Rhee & Yutaka Shimomoto, 1999 \$35.00 (paperback)
- 9. Rising to the Challenge in Asia: A Study of Financial Markets
  - Asian Development Bank
  - Vol. 1: An Overview, 2000 \$20.00 (paperback)
  - Vol. 2: Special Issues, 1999 \$15.00 (paperback)
  - Vol. 3: Sound Practices, 2000 \$25.00 (paperback) Vol. 4: People's Republic of China, 1999 \$20.00
  - (paperback)
  - Vol. 5: India, 1999 \$30.00 (paperback)
  - Vol. 6: Indonesia, 1999 \$30.00 (paperback)
  - Vol. 7: Republic of Korea, 1999 \$30.00 (paperback)
  - Vol. 8: Malaysia, 1999 \$20.00 (paperback)
  - Vol. 9: Pakistan, 1999 \$30.00 (paperback)
  - Vol. 10: Philippines, 1999 \$30.00 (paperback)
  - Vol. 11: Thailand, *1999 \$30.00 (paperback)* Vol. 12: Socialist Republic of Viet Nam, *1999 \$30.00*
- (paperback)10. Corporate Governance and Finance in East Asia: A Study of Indonesia, Republic of Korea, Malaysia,
  - Philippines and Thailand J. Zhuang, David Edwards, D. Webb, & Ma. Virginita

J. Zhuang, Davia Eawaras, D. weoo, & Ma. virginita Capulong Vol. 1: A Consolidated Report, 2000 \$10.00 (paperback)
Vol. 2: Country Studies, 2001 \$15.00 (paperback)
11. Financial Management and Governance Issues
Asian Development Bank, 2000
Cambodia \$10.00 (paperback)
People's Republic of China \$10.00 (paperback)
Mongolia \$10.00 (paperback)

- Pakistan \$10.00 (paperback)
- Papua New Guinea \$10.00 (paperback)
- Uzbekistan \$10.00 (paperback)
- Viet Nam \$10.00 (paperback)
- Selected Developing Member Countries \$10.00 (paperback)
- Government Bond Market Development in Asia Edited by Yun-Hwan Kim, 2001 \$25.00 (paperback)
- Intergovernmental Fiscal Transfers in Asia: Current Practice and Challenges for the Future Edited by Paul Smoke and Yun-Hwan Kim, 2002 \$15.00 (paperback)
- Guidelines for the Economic Analysis of Projects Asian Development Bank, 1997 \$10.00 (paperback)
- Guidelines for the Economic Analysis of Telecommunications Projects Asian Development Bank, 1997 \$10.00 (paperback)
- Handbook for the Economic Analysis of Water Supply Projects Asian Development Bank, 1999 \$10.00 (hardbound)
- Handbook for the Economic Analysis of Health Sector Projects Asian Development Bank, 2000 \$10.00 (paperback)
- Handbook for Integrating Povery Impact Assessment in the Economic Analysis of Projects Asian Development Bank, 2001 \$10.00 (paperback)
- Handbook for Integrating Risk Analysis in the Economic Analysis of Projects Asian Development Bank, 2002 \$10.00 (paperback)
- Handbook on Environment Statistics Asian Development Bank, 2002 \$10.00 (hardback)
- Defining an Agenda for Poverty Reduction, Volume 1 Edited by Christopher Edmonds and Sara Medina, 2002 \$15.00 (paperback)
- Defining an Agenda for Poverty Reduction, Volume 2 Edited by Isabel Ortiz, 2002 \$15.00 (paperback)
- Economic Analysis of Policy-based Operations: Key Dimensions Asian Development Bank, 2003 \$10.00 (paperback)

## Institutions and Policies for Growth and Poverty Reduction: The Role of Private Sector Development

Rana Hasan, Devashish Mitra, and Mehmet Ulubasoglu examine the relationship between poverty, economic growth, and indicators of business regulations and governance. The evidence suggests that regulations that facilitate the creation of new enterprises, and governance indicators that capture conduciveness to economic activity as distinct from political freedom, can play an important role in promoting growth and reducing poverty.

#### About the Asian Development Bank

The work of the Asian Development Bank (ADB) is aimed at improving the welfare of the people in Asia and the Pacific, particularly the 1.9 billion who live on less than \$2 a day. Despite many success stories, Asia and the Pacific remains home to two thirds of the world's poor. ADB is a multilateral development finance institution owned by 64 members, 46 from the region and 18 from other parts of the globe. ADB's vision is a region free of poverty. Its mission is to help its developing member countries reduce poverty and improve the quality of life of their citizens.

ADB's main instruments for providing help to its developing member countries are policy dialogue, loans, technical assistance, grants, guarantees, and equity investments. ADB's annual lending volume is typically about \$6 billion, with technical assistance usually totaling about \$180 million a year.

ADB's headquarters is in Manila. It has 26 offices around the world and has more than 2,000 employees from over 50 countries.

Asian Development Bank 6 ADB Avenue, Mandaluyong City 1550 Metro Manila, Philippines www.adb.org/economics ISSN: 1655-5252 Publication Stock No.